Ethical thinking in a disciplinary context: the ethical development of undergraduates and expectations of tutors in the arts, social and pure sciences

Dissertation submitted in accordance with the requirements of the University of Chester for the degree of Master of Arts

Dr Ruth L Healey

September 2012

Acknowledgements

I would like to thank the students and tutors in English, Geography, and Animal Behaviour and Welfare from the case study university who participated in this research. Without them, this research would not have been possible. I would also like to acknowledge the support of Dr Chris Ribchester and Kimberley Ross, with whom a previous related project was developed. It was this work which inspired this research. Finally, I would like to thank Dr Julie-Anne Regan for her insightful, detailed and supportive comments in preparation of this dissertation.

Abstract

Barnett (2000) argues that universities need to prepare students for 'supercomplexity', where "the very frameworks by which we orientate ourselves to the world are themselves contested" (p. 257). Learning to think through ethical issues develops critical thinking skills for dealing with supercomplexity, since the frameworks the students use to consider ethical issues are contested and likely to change. Yet, Boyd *et al.* (2008) question whether universities actually produce graduates who are prepared "for practical and ethical engagement with their scholarly, professional and personal worlds" (p. 38). Moreover, we might expect differences in ethical thinking between disciplines given that the nature of ethical issues studied varies by discipline.

The overall aim of this research was to explore the development of undergraduates' ethical thinking during their programmes and to compare how it aligns with the expectations of their tutors and to discuss the implications for teaching and learning ethics in higher education. To address this aim the research objectives were to assess whether the ethical development of undergraduate students varies by discipline, gender and year; to analyse how the nature of ethical thinking expected by tutors varies between disciplines and evaluate the extent to which this aligns with the students' ethical development; and to discuss the implications for enhancing the teaching and learning of ethics. Most emphasis is placed on the first objective. To address these objectives, a questionnaire exploring students' ethical understandings and level of ethical development, was given to students in all three undergraduate years of the English (art), Geography (social science) and Animal Behaviour and Welfare (pure science) programmes at an English University. In total 335 students responded. Interviews were then conducted with tutors teaching on the three programmes discussing the nature of ethics within their disciplines, how ethics was taught and what ethical thinking skills they wanted their students to develop.

The key findings are that: 1) There are no significant differences between disciplines in terms of student ethical development. 2) There is some evidence of differences between years, but there was not clear evidence of progression over the three years of the undergraduate programme. 3) Male students demonstrate less ethical development than their female counterparts. 4) Tutors across all three disciplines have similar expectations in terms of the nature of ethical thinking desired. 5) Most of the students exhibit lower levels of ethical development than their tutors expected. It is suggested the skill of 'ethical thinking' should be included in programme outcomes and that teaching and learning strategies which cast students in the role of active, social and creative learners offer the best potential to enhance student ethical thinking abilities.

Declaration

This work is original and has not been previously submitted in support of a Degree, qualification or other course.

Signed

Date

Contents

Acknow	wledgements	II	
Abstra	ct	III	
Declar	ation	IV	
List of	Tables	VII	
List of	Figures	VIII	
Chapt	er 1	1	
Introd	luction: The importance of ethics		
Chapt	er 2	7	
Ethica	I thinking: discipline, year and gender variations		
2.1	Defining ethical thinking and meta-ethical development	7	
2.2	Ethical thinking in higher education	8	
2.3	Arts, social and pure science discipline comparison	11	
	2.3.1 English: Arts	11	
	2.3.2 Geography: Social science	12	
	2.3.3 Animal Behaviour and Welfare: Pure science	12	
2.4	Progression by year	13	
2.5	Variation by gender	15	
2.6	Implications	16	
Chapt	er 3	18	
Methe	odology: A comparative case study approach		
3.1	Rationale and approach	18	
3.2	Data collection, recruitment and analysis	20	
	3.2.1 Student questionnaire	21	
	3.2.2 Tutor interviews	25	
Chapt	er 4	28	
Stude	nt ethical development by discipline, year and gender		
4.1	Characteristics of respondents	28	
4.2	Meta-ethical questionnaire	30	
	4.2.1 Discipline analysis	32	
	4.2.2 Year group analysis	36	
	4.2.3 Gender analysis	40	
4.3	Meta-ethical questionnaire elements	44	
4.4	Open ended questions	47	
	4.4.1 What is ethics?	47	
	4.4.2 Example of ethical issue previously studied	52	
	4.4.3 Example of ethical decision	55	
4.5	Summary	59	
Chapt	er 5	60	
Tutor	Tutor reflections on the nature of ethical thinking within the disciplines		

Chap	Chapter 6		
Conc	lusions		
6.1	Implications of findings for teaching and learning	71	
6.2	Limitations and reflections	76	
6.3	Future research	76	
Refe	rences	78	
Арре	endices	81	
1	Ethical thinking in a disciplinary context questionnaire	81	
2	Participant information sheet: ethical thinking in a disciplinary context interviews with tutor	89	
3	Consent form to participate	90	
4	Analysis of 'Element I: source and type of moral answers' by discipline, year and gender	91	
5	Analysis of 'Element II: role of authority' by discipline, year and gender	93	
6	Analysis of 'Element III: nature of multiplicity' by discipline, year and gender	95	
7	Analysis of 'Element IV: personal responsibility and relationship with multiplicity' by discipline, year and gender	97	
8	Analysis of 'Element V: purpose of moral discussions' by discipline, year and gender	99	

List of Tables

Table		Page
2.1	Outline of the nine positions of ethical development proposed by Perry (1999)	14
3.1	Questionnaire ethical considerations	23
3.2	Meta-ethical questionnaire response options	25
3.3	Interview ethical considerations	26
4.1	Questionnaire responses by programme	29
4.2	Responses by year	30
4.3	Responses by gender	30
4.4	MEQ usable responses by year	31
4.5	MEQ usable responses by gender	31
4.6	Mean ethical score and range by discipline	34
4.7	Summary of characteristics of different types of ethical development	34
4.8	Mean ethical score by year	37
4.9	Mean ethical score and range by year	37
4.10	Mean ethical score by gender	41
4.11	Mean ethical score and range by gender	41
4.12	Statistical differences between elements	45
4.13	Statistical analysis of elements by subject, year and gender	46
4.14	What is ethics? Categories and examples	47
4.15	Example of ethical issue previously studied by the students: Categories and examples	52
4.16	Example of ethical decision: Categories and examples	55
4.17	Research question key findings	59
5.1	Expected ethical thinking in relation to programme outcomes and student development	67
6.1	Further research questions	77

List of Figures

Figure		Page
4.1	Mean ethical score by discipline	33
4.2	A 'box and whiskers' plot on the ethical scores by discipline	33
4.3	Participant ethical scores by different types of ethical development and discipline	35
4.4	Overall year group differences	37
4.5	A 'box and whiskers' plot of the ethical scores by year	38
4.6	Mean ethical score by year	39
4.7	Participant ethical scores by different types of ethical development and year	40
4.8	Mean ethical score by gender	41
4.9	A 'box and whiskers' plot of the ethical scores by gender	42
4.10	Mean ethical score by gender for each discipline	42
4.11	Participant ethical scores by different types of ethical development and gender	43
4.12	The mean ethical score per element	44
4.13	What is ethics? Percentage of responses in each category by discipline, year and gender	51
4.14	Example of ethical issue previously studied: Percentage of responses in each category by discipline, year and gender	54
4.15	Example of ethical decision: Percentage of responses in each category by discipline, year and gender	58

Introduction: The importance of ethics

"Educators need to give greater attention to the teaching of ... ethics as part of our contribution to the education of responsible citizens." (Hay & Foley 1998: 169)

As competition within the education market has increased, universities have progressively attempted to define the distinctive characteristics of their graduates (Barrie 2004; 2006; 2007). Barrie (2004) has identified 'Ethical, Social and Professional Understanding' as one of five key graduate attributes. This attribute means that "graduates of the university will hold personal values and beliefs consistent with their role as responsible members of local, national, international and professional communities" (Barrie 2004: 270). This graduate attribute relates to the need to prepare students for 'supercomplexity', where "the very frameworks by which we orientate ourselves to the world are themselves contested" (Barnett 2000: 257). As students prepare for careers in the 21st Century they require global competences to understand the complex world in which they live (Bartell 2003). Healey et al. (2011) argue that learning to think through ethical issues develops critical thinking skills for dealing with supercomplexity. Ethical issues are an example of supercomplexity, as the frameworks the students use to consider ethical issues are both contested and likely to change. In increasingly dynamic professional and social lives, graduates need these skills to enable them to negotiate an uncertain world. Yet, Boyd et al. (2008: 38) question whether graduates are leaving university prepared "for practical and ethical engagement with their scholarly, professional and personal worlds."

This project came about in response to several experiences in which many of the students I teach demonstrated a lack of consideration of ethics within their studies. One such incident occurred when supervising a physical geography field trip. The students were asked to analyse water quality by indexing the number of different animal species they had collected in their sample. After they had noted the species in their sample several students did not think twice about returning the creatures to the water by tipping the trays upside down. When questioned about their actions some of the students recognised how their actions were inappropriate as the sudden upturning of the tray disturbs the organisms more than immersing the tray in water to allow them to leave the tray on their own. The students did not appear to reflect or think critically about how they would deal with situations similar to this in the future. Experiences such as this led me, with two colleagues, to propose a Learning and Teaching Institute (LTI) project in the Department of Geography and Development Studies (2010-11) which explored the ethical development of students in the department in different years and how the teaching of ethics, particularly in the second year, might be enhanced (Healey et al. 2011). This project led me to contemplate the extent to which the findings for geography related to other disciplines.

Understanding and learning about ethics and ethical issues is a skill that develops throughout an individual's life (Knapper & Cropley 2000). What happens within an educational environment is only part of the ethical learning that individuals experience (Kuh 1994). Despite this, higher education has an important role to play

2

in enabling students to recognise and understand ethical issues (Beck & Murphy 1994; Cortese 2003).

Within higher education the nature of the ethical issues studied by graduates varies between disciplines (Lane & Schaupp 1989; Rooy & Pollard 2002). For example, the ethical issues pure scientists face when testing on human subjects or undertaking animal experiments are of a different nature from those dealt with by social scientists when interviewing or observing people, or those explored in literature when deciding whether a character made the appropriate ethical choice. However, in terms of critical thinking, many ethical issues are multidisciplinary in nature, for example assisted suicide may be studied from many different disciplinary perspectives, yet the ways in which students might approach and think about such a topic may differ between disciplines. For example, science students may analyse the issue from the perspective of the medical issues of the individual body, whereas social scientists may consider the implications of assisted suicide for broader society.

For disciplines which involve primary research with animals or people, for many students their main contact with ethics relates primarily to the ethics of undertaking research (Boyd *et al.* 2008). Students are carefully guided through the process of seeking ethical approval for projects, especially students on accredited courses which have prescribed ethical standards (e.g. British Psychological Society accredited courses). Going through ethical clearance procedures has in many cases become relatively mechanistic, after which students may give ethics little further consideration. However, in terms of a graduate attribute, ethics is more concerned with developing individuals to have the broader skill of *thinking* ethically in all parts of their lives, not just in research.

This research was conducted in an English University. The institution became a university in 2005 with an emphasis on teaching influenced by research. In its mission statement, the University identifies four key aspects of the education students receive at the institution:

- the pursuit of learning for its own sake
- the development of skills relevant for the needs of a healthy society
- the encouragement of students' character and values
- the importance of community in the learning enterprise

Additionally, the University's Learning and Teaching Strategy refers to the development of curricula that 'encourage reflective engagement with community and society' (University [in England] 2010: no page). Engaging students with ethical thinking supports the development of the character and values of graduates who have the skills to contribute to a healthy society.

As already noted this project builds upon a previous study by the author (Healey *et al.* 2011). In this project all students studying in the Department of Geography and Development Studies were asked to complete a questionnaire about their ethical development. This covered four programmes: Single Honours Geography, Combined Honours Geography, Combined Honours Natural Hazard Management and Combined Honours International Development Studies. In total 198 questionnaires were completed across all three years. Alongside this, interviews were conducted with thirteen tutors about their views on the importance of ethics

and what ethical awareness and skills they expected students to have by the time they graduate. The current study extracted the data from the earlier project for those students identified as studying one of the social science programmes in the department. By asking the same questions in an arts discipline and a pure science discipline this research was able to test the hypothesis that the ethical development of students varies by discipline.

The overall aim of this research was to explore the development of undergraduates' ethical thinking during their programmes and to compare how it aligns with the expectations of their tutors and to discuss the implications for teaching and learning ethics in higher education. This project has five main objectives:

- 1. To analyse the ethical development of students in three academic programmes in the arts, social and pure sciences.
- 2. To investigate the extent to which there is progression in the ethical development of students in different years across the three academic programmes.
- 3. To examine the ethical development of students' ethical thinking by gender across the three academic programmes.
- To analyse how the nature of ethical thinking expected by tutors varies between disciplines and evaluate the extent to which this aligns with the students' ethical development.
- 5. To discuss the implications of the findings for enhancing the teaching and learning of ethics.

Most emphasis is placed on the first three objectives. This contributes to the research field in four main ways: 1) by comparing student ethical development between disciplines; 2) by comparing ethical development within disciplines by gender and year; 3) by replicating Clarkeburn *et al.*'s study (2003) in a different university and with different disciplines; 4) by assessing the extent to which student ethical development aligns with tutor expectations.

The following chapter reviews the literature on teaching ethics in higher education and defines what is meant by ethical thinking and meta-ethical development, before exploring the differences in ethical development between disciplines, year of study and gender. Chapter 3 explains the mixed-method approach adopted to explore ethical thinking and ethical development at the case study university. The analysis is then split between two chapters. Chapter 4 discusses student ethical development by discipline, year and gender, whilst Chapter 5 discusses tutor expectations about ethical thinking within the three disciplines in relation to student ethical development. Chapter 6 concludes the dissertation by reflecting upon the findings and discussing their implications for teaching ethics in higher education.

Chapter 2

Ethical thinking: discipline, year and gender variations

2.1 Defining ethical thinking and meta-ethical development

Ethics is a commonly-used label concerning a complex set of concepts and behaviours (Israel & Hay 2006). Ethics are distinct from morals which are concerned with the individual's personal character, whereas ethics focuses upon the broader social system in which morals are applied. Hence a sophisticated understanding of ethics recognises not just an individual's morals but how these are a part of the broader social system, influencing people's behaviour, and how people have different morals and ethical perspectives.

This project is interested in how 'ethical reality' is constructed by students. It adopts Clarkeburn *et al.*'s (2003) concept of 'meta-ethical development' which describes how students construct ethical realities, for example how students interpret the nature of ethical properties, attitudes and judgements. This notion is complemented by the concept of 'ethical thinking'. How students construct ethical reality influences their ability to think ethically. For the purposes of this research 'ethical thinking' encompasses two elements (Clarkeburn *et al.* 2002):

- 1. **Ethical sensitivity**: an ability to perceive the ethical implications of a situation. Without the initial recognition of moral facts alongside scientific or 'hard' facts it is not possible to make moral decisions. Ethical sensitivity is also about an ability to understand the moral networks and implications of moral actions.
- 2. **Moral reasoning**: an ability to engage in sound moral reasoning and use practical problem solving strategies. To make a judgement about which course of action is morally right (or fair, just, morally good or adequate) and thus label one possible line of action as what a person ought (morally) to do in that situation. Moral reasoning is also called 'moral cognitive skills'.

In order to discuss ethical issues, students require decision making skills, skills to understand and recognise moral issues, and skills to consider and decide upon solutions to moral problems (Clarkeburn *et al.* 2002). These elements of ethical thinking are contextualised within the broader skill of critical thinking:

"Ethical learning is impossible without the development of critical reasoning (Kant, 2003) and, at the same time, critical reasoning is reinforced by the aspiration for justice and the independence sought by ethical learning" (Boni & Lozano 2007: 825).

Meta-ethical development therefore occurs through the improvement of critical thinking skills in relation to ethical issues. Ethical thinking is a particular type of critical thinking. Wolf *et al.* (2010) argues that critical thinking skills can "give students the tools to understand what they are learning" (p. 43). This ability relates to analytical, interpretation, inference, explanation, and evaluation skills (Facione, 2000). Ethical thinking requires these skills, enabling students to monitor and, where appropriate, correct their own moral reasoning; meaning that critical ethical thinking is about "judging in a reflective way what to do or what to believe" (Facione, 2000: 61). Students who have limited meta-ethical development consider reality to be certain, and believe in absolute answers (Clarkeburn *et al.* 2003). Learning to think ethically, to critically reflect on ethical issues, offers the opportunity for students to develop their understanding of ethical reality, recognising the complexities and uncertainties within life.

2.2 Ethical thinking in Higher Education

Over the last decade public trust, in business and more recently in politicians, has eroded (Gao *et al.* 2008; Ruhe & Lee 2008; Carrell 2009). It is increasingly

8

recognised that ethically and socially responsible behaviour plays a crucial role in good business practice (Nicholson & DeMoss 2009) and that "moral meaning and agency are fundamental to the definition of professions" (Robinson 2005: 2). Hence, graduate careers require more than just professional competence; they need to include a moral dimension (Boni & Lozano 2007).

Ethical thinking is a part of professional responsibility (Solbrekke & Karseth 2006). In the wake of policy led attempts to 'professionalise' aspects of academic practice (for example, Higher Education Academy 2006) there has been a renewed interest in the values that define academic life (Macfarlane & Cheug 2008). Some have gone as far as to ask whether or not society's expectations of higher education should be codified, and does higher education need a Hippocratic Oath (Watson 2007). The Association of Masters of Business Administration (Association of MBAs) has even introduced an MBA oath which has been pledged by 7,142 students from business schools across the world (Matthews 2012). Higher Education institutions should be leaders in the development of the cultures and the societies in which they are situated (Walesh 2012). It is within the environment of universities where students may first critically discuss the realities of citizenship and test its moral boundaries (Bruhn 2008), exploring the nature of social responsibility (Vujakovic & Bullard 2001) and developing the skills which optimistically will contribute towards transforming society for the better (Wellens et al. 2006).

Hargreaves (2008) argues that higher education in the UK aims to develop the intelligence and critical thinking skills of undergraduates. Ethical thinking is one

element of this (Hay & Foley 1998; Smith 1995). It is important to recognise that students are already ethically developed to varying degrees, yet higher education should ensure that students graduate with these skills. However, Escámez *et al.* (2008) found that current ethical teaching "often left students unarmed to cope with the frequent conflicts between ends, responsibilities, rights and duties that are bound to occur in their professional careers" (p. 43). The experiences students are having in Higher Education do not always prepare them for the potential moral questions they need to respond to in their post-graduation employment. Further knowledge as to factors which influence the nature of students' meta-ethical development offers opportunities to address these shortfalls.

The role of higher education should be to develop an ethics education which emphasises the significance of ethical consciousness in autonomous individuals (Hay 1998) rather than one structured around a set of 'rules' for moral behaviour (Hay & Foley 1998; Clarkeburn *et al.* 2002). This form of teaching offers individuals support to become ethically accountable for their own choices and actions whilst situating them within a supportive ethical community. The skill to think ethically is one of the most important "generic skills that future graduates should have" (Escámez *et al.* 2008: 50). In developing skills which allow them to handle the moral issues associated with the real world students are also better prepared for employment (Hay & Foley 1998). However, the extent to which ethics is seen as important in the discipline may vary.

2.3 Arts, social and pure science discipline comparison

Society's present problems are global and multidisciplinary in nature (Boni & Lozano 2007). Issues such as pollution, human rights, the fight against poverty, world security and so on, involve everybody and require multi and inter disciplinary approaches (Boni & Lozano 2007). Within higher education there are also areas of academic concern which cross disciplines, for example academic dishonesty, plagiarism, collusion and cheating (Ellery 2008; Colnerud & Rosander 2009). Yet, previous work has found significant differences between disciplines in terms of ethical beliefs (Lane & Schaupp 1989). This project considers ethics across three programmes: English (English literature), Geography (Human Geography, International Development Studies, and Natural Hazard Management) and Biosciences (Animal Behaviour). These discipline areas cover respectively the arts, social and pure sciences.

2.3.1 English: Arts

The discipline of English contains significant opportunities for exploration of ethical issues. For example, in English literature stories have the power to "train the moral imagination" (Hilder 2005: 42); in English language, research with participants raises ethical issues around working with participants in an ethically sensitive manner; and in creative writing issues of representation highlight ethical concerns. Yet, the English subject benchmark statement has no mention of ethics in relation to the discipline (QAA 2007a). However, for the sub-discipline of creative writing

and English language the importance of ethics is noted as a cognitive ability (NAWE 2008) and as an approach to research (HEA 2011).

2.3.2 Geography: Social science

Smith (1995) argues that moral issues are often marginalised within contemporary education, and that the discipline of geography is particularly well positioned to address this deficiency. Geography deals with many "inherently controversial subjects, from population control to environmental change" (Vujakovic & Bullard 2001: 276), providing a significant range of contemporary topics in which to situate ethical discussion. For example, 'sustainable development', a contested concept which underpins many contemporary geographical debates, is replete with ethical questions. The geography benchmark statement claims that:

"Geography fosters a range of personal attributes relevant to the world beyond HE, which will promote geographers' ability to engage in lifelong learning, to consider ethics and values, and to contribute to the wider community" (QAA 2007b: 3).

The benchmark statement emphasises research and field based studies in relation to ethics, but also recognises "the moral and ethical issues involved in debates and enquiries" within the discipline (QAA 2007b: 5).

2.3.3 Animal Behaviour and Welfare: Pure science

Animal Behaviour and Welfare is categorised as a bioscience. Bioscientists face numerous ethical considerations whether it is choosing where to apply for funding, the research topic itself, or their interaction with animal (and sometimes human) research subjects (Clarkeburn *et al.* 2002). However, despite the recognition of the importance of ethics, the extent to which it is taught explicitly within the life sciences varies significantly (Clarkeburn *et al.* 2002). This may be because members of staff are concerned that "ethics would demand too much time in a curriculum at a cost to the 'core' scientific subjects" (Clarkeburn *et al.* 2002: 66). Yet with science increasingly being taught within a social context (Reiss 1999), the Biosciences subject benchmark statement explicitly mentions ethics nine times in relation to critical assessment of intellectual arguments, professional codes of conduct, research methods, and a need to interpret decisions in relation to the broader social context (QAA 2007c).

2.4 Progression by year

Undergraduate higher education programmes aim to develop their students' analytical and critical thinking skills (Moon 2008). The curriculum is designed to support students at different stages in their development, and as such it would be expected that students' skills improve as they progress through their degrees. Perry's (1999) model of intellectual and ethical development assumes that students are at different stages of ethical development at different points in their university studies, with the expectation that the further they go through the academic system the greater their ethical engagement. This argument is supported by research with pre-service teacher education students, where the findings indicated moderate to significant changes in student moral judgement and reasoning (Reiman 2002). Clarkeburn *et al.* (2003) used Perry's (1999) scheme to categorise different students' ethical development (Table 2.1). Students are considered to be more ethically developed when they achieve commitment characteristics (Type C). However, research into changing ethical perceptions and understanding found there to be no significant differences by year (Dellaportas 2002; Ludlum & Mascaloinov 2004). Ludlum & Mascaloinov (2004) suggest that this may relate to the thesis that education does not influence educational beliefs. Yet, continue by pointing out that the lack of significance in their study may relate to the small proportions of students in the later years of their degree who participated in the study (Ludlum & Mascaloinov 2004).

Туре	Position	Short description
	1. Basic Duality	Reality with no uncertainty or diversity, belief in absolute answers and authorities, obedience instead of responsibility
A	2. Multiplicity pre-legitimate	Perception of diversity, but classified as unwarranted confusion
	3. Multiplicity sub-ordinate	Diversity and uncertainty accepted, but classified as temporary in a search for absolute answers
В	4. Multiplicity correlate or relativism subordinate	First attempt to accommodate diversity and uncertainty with the expectation to do the right thing either by accepting no answers as truth-bearing or by subordinating to authority
	5. Relativism correlate, competing, diffuse	Revolution through the inability to assimilate uncertainty and the existence of absolute answers, everything becomes relativistic
	6. Commitment foreseen	Awareness of the need of orientation in a relativistic world, experimentation with different methods
С	7. Initial commitment	Initial commitment in one area.
	8. Orientation in implications of commitment	Experience with commitment and responsibility, time of personal choices of how to fulfil perceived responsibility
	9. Developing commitment	Affirmation of identity among multiple responsibilities, commitment seen as an on-going activity

Table 2.1: Outline of the nine positions of ethical development proposed by Perry (1999)

Source: Clarkeburn et al. (2003: 446)

Although it might be expected that student's meta-ethical development might increase as they progress through their degrees, the relationship between skill development and ethical development may not be direct. Rather, greater experience of dealing with real life ethical issues as people progress through life may have a greater impact upon an individual's engagement with the complexity and contingent nature of ethical issues.

2.5 Variation by gender

Numerous published studies, using a range of tools, have found that female students tend to be more ethical than their male counterparts (Barnett & Brown 1994; Donoho et al. 2012; Persons 2009; Tse & Au 1997; Whipple & Swords 1992). It is argued that women may be predisposed to support ethical positions more strongly through their conditioning as caregivers (Ludlum & Mascaloinov 2004). However, further research indicates some anomalies and nuances in this general trend. For example, some studies have found no significant difference between men and women when investigating their perceptions about ethics (Kelger 2011). These differences may relate to the challenges of measuring what it means to be ethical. Alongside this, Phau & Kea (2007) found that males were generally more ethical than females in research in Singapore, Hong Kong and Australia. As many of the studies that have found females to be more ethical than men were conducted in North America, then these differences might relate to cultural upbringing and background. However, even in the US where women have been found to be slightly more ethical than men, it is acknowledged that these differences were only very slight and in the same direction for both genders (Ludlum & Mascaloinov's 2004). Furthermore, Peterson et al. (2001) found that the ethical views of men tended to become more in line with women's as they aged.

To summarise, although the majority of research has found female students, on a variety of different measurements, to be more ethical than their male counterparts, there are definitive exceptions. The range of ways in which ethicality is measured means that it is not possible at this point to conclude that there is a gender difference.

2.6 Implications

This project emerged from a concern over a lack of ethical awareness and consideration by many students known to the author. This chapter has considered the factors that might influence student ethical development, demonstrating the potential for variety in ethical development by discipline, year and gender. In order to explore the similarities and differences in the requirements of ethical thinking within these disciplines it is important firstly to recognise the ethical knowledge students from different disciplines already have. An awareness of students' current ethical development provides a basis from which to interpret the ethical skill development needs in a discipline, by year and gender. Secondly, it is necessary to understand the current level of meta-ethical development in different student cohorts. Without this it is not possible to design teaching to support ethical progression (Clarkeburn et al. 2003). Thirdly, it is important to recognise how student's development may vary by gender. A greater understanding of variability, or lack of variability, of ethical development by gender offers the opportunity to consider designing curricula to support the range of different levels of ethical development within a cohort. Greater knowledge as to the factors that influence, or do not influence, the ethical development of students offers opportunities to

16

enhance the teaching and learning of ethics in higher education. The next chapter discusses the methodology of the research and explaining how the aims of the research are addressed.

Methodology: A comparative case study approach

3.1 Rationale and approach

This research takes the position that "the social world can only be understood from the standpoint of the individuals who are part of the on-going action being investigated" (Cohen *et al.* 2000: 19). This follows the belief that reality is the result of individual cognition. In order to analyse students' ethical development it is necessary for them to 'voice' their views as to the nature of ethics. This research adopts a comparative case study approach by comparing selected programmes from three contrasting departments: English, Geography and Animal Behaviour and Welfare. Case studies are generally the preferred approach when "'how' or 'why' questions are being posed, when the investigator has little control over events, and when the focus is on a contemporary phenomenon within some real-life context" (Yin 2002: 1). This research is asking 'how' much understanding students have of ethics within their different disciplines, making a case study approach appropriate. Case study research "aims to explore and depict a setting with a view to advancing understanding of it" (Cousin 2005: 421). Case study research originates from cultural anthropology in which the research site is considered as the 'field' and the researcher, often as a participant observer, gathers and analyses data from this location over a period of time (Cousin 2009). Following Bassey's (1999) caution, this case study aspires to make 'fuzzy generalisations' which assert that everything

18

is a matter of degree, and nothing is certain. Therefore the generalisations will be predicted in terms of 'may' rather than 'will' (Cousin 2005; 2009).

Before discussing the method in detail it is important to acknowledge the author's positionality in relation to the research. Who the researcher is and their relationship with the research has both positive and negative influences on how the research is constructed and conducted, and how the findings are interpreted. There are three main parts of my identity which are important here: 1) my position as a lecturer, 2) my own moral framework, and 3) my motivation for undertaking the study. My positionality as a lecturer at the case study university provided me with access to gatekeepers in different departments and potentially put me in a position of power in relation to the students who participated in the research. As a colleague at the same institution some of the gatekeepers in other departments, may have been more willing to enable access to their students, than they would have been to someone from a different institution or an undergraduate student requesting access. As a lecturer, and therefore a perceived authority figure, the students, even though not taught by me, may have felt obliged to complete the questionnaire. Secondly, my own moral framework influences my interpretation of ethical development. For example, this has influenced how I found Perry's classifications and the Meta-Ethical Questionnaire an acceptable interpretation, and tool, respectively for understanding ethical development. Finally, the motivation to research this topic in this particular way influences the study. This dissertation was conducted as part of my career development, with the desire to develop my quantitative research skills. The choice of the topic, as argued in the

19

introduction, was motivated with the desire to enhance student understanding of ethics in higher education. By acknowledging the influence of my positionality it is possible to recognise how the potential positive and negative impacts on the research balance out.

The remainder of this chapter discusses the methods used to address the five main

research questions:

- 1. What are the differences and similarities in the ethical development of students in three academic programmes in the arts, social and pure sciences?
- 2. What is the extent to which there is progression in the ethical development of students in different years across the three academic programmes?
- 3. Is there any difference in the ethical development of students by gender across the three academic programmes?
- 4. What are the differences and similarities in the nature of the ethical thinking expected by tutors between the three academic programmes and to what extent does this align with the students' ethical development?
- 5. What are the implications of the findings for enhancing the teaching and learning of ethics?

3.2 Data collection, recruitment and analysis

To address the research questions, a mixture of qualitative and quantitative methods were used to consider the "multiplicity of meanings, representations and practices" (Smith 2001: 24) of ethics in higher education. This includes two methods of data collection: 1) an anonymous questionnaire with students; and 2) semi-structured interviews with four tutors (in addition to the thirteen already undertaken in Geography). In order to maintain a common institutional context, this research took place in different departments at the same institution.

3.2.1 Student questionnaire

The first three research questions were addressed through the use of a There were three reasons for choosing a self-completion questionnaire. questionnaire to collect both qualitative and quantitative information. Firstly, the research aimed to compare findings with the study by Clarkeburn et al. (2003). This study had used a questionnaire to gather data to provide students with an 'ethical score' to identify their level of meta-ethical development. To ensure consistency the same tool was used for this research. Secondly, alongside qualitative differences between programmes, a questionnaire offers the opportunity to look for statistically significant trends and differences. A questionnaire enables responses to be gathered from large numbers relatively quickly, and cost efficiently (De Vaus 1991; Fink 1995). Thirdly, when discussing a topic such as ethics, different students' development varies in relation to their context and belief systems. Consequently a qualitative method of data collection would enable participants to express their views in their own words. An anonymous open-ended questionnaire is believed to be an appropriate tool for enabling students to explain in their own words what the topic means to them. A one-to-one interview, or focus group, on the topic of ethics, may intimidate students and lead them to feel uncomfortable (Miller 1999). However, a questionnaire is likely to produce less in-depth responses, and prevent the researcher from probing responses further.

The questionnaire explored student meta-ethical development by incorporating Clarkeburn *et al.*'s (2003) Meta-ethical Questionnaire (MEQ) (see Appendix 1 for full questionnaire). The MEQ assessed how students constructed ethical reality,

21

exploring how they interpret the nature of ethical properties, attitudes and judgements. As argued in Chapter 2 meta-ethical development occurs through the improvement of ethical thinking skills. The first section of the questionnaire offered students the opportunity to express their understanding of ethics through three open-ended questions. These questions enabled the students to explain their interpretation of ethics in their own words by explaining the term, providing an example from their past education, and identifying a time when they have put ethics into action in some way. The second section of the questionnaire was quantitative asking students to choose their responses along a scale: the MEQ. This questionnaire was rigorously developed and tested with a cohort of 478 Life Studies students at the University of Glasgow and the findings published in *Studies in Higher Education*.

After contacting programme leaders in the Departments of English and Biological Sciences the researcher was given permission to work with their students. During the first two weeks of the academic year all students, at each level, studying on the programmes in English and Animal Behaviour and Welfare were asked to complete the questionnaire already used in the Department of Geography and Development Studies. In order to encourage a reasonable response rate, the questionnaires were given out in teaching sessions and the tutors made time for the questionnaires to be completed then and there (see Table 3.1 for ethical considerations). So that all students had the opportunity to complete the questionnaire an electronic version was also made available (though none were received electronically).

Table 3.1: Question	naire ethical	considerations
---------------------	---------------	----------------

Ethical issue	Questionnaire
(Source: Hay 2003)	
Access	 Permission was sought from programme leaders in each of the
	departments to work with their students.
Consent	• Students were invited to take part in the questionnaire at the beginning or end of a teaching session they had attended.
	 It was made clear in a spoken introduction that participation was voluntary.
	• A full Participant Information Sheet (PIS) was attached to the
	questionnaire (Appendix 1). In order to minimise disruption to the
	teaching session this was presented at the same time as the questionnaire.
Confidentiality	 All questionnaire responses were anonymous.
	• The information collected could not be traced back to an individual student.
Harm	• The PIS made clear that there would be no adverse effects or benefits to
	the students whether they chose to take part in the study or not.
	• The power relations of being a lecturer were reduced by working with students who I do not teach in English and Animal Behaviour and
	Welfare. In the previous year when data was collected with Geography
	students, a research assistant who did not teach the students organised the collection of data from the students.
Cultural awareness	• The topic of ethics is sensitive with moral attitudes and perspectives being variable between different people and cultures. The questionnaire was phrased in such a way that students could identify how they perceive
	ethics.
Dissemination	• An electronic version of this dissertation will be uploaded to my staff
	website so that participants may read the findings.
	• An email of the abstract will be sent around the programmes where the
	questionnaires were collected with a link to the electronic final report.

The questionnaire data was analysed in two main ways: 1) within the discipline for relationships between characteristics of the participants (year and gender) and ethical development; 2) between the disciplines, year group and genders for ethical development. This analysis involves two main aspects: firstly, an interpretation of the different ways in which students understood ethics. The analysis of these openended responses took a grounded approach. The responses were all read to identify key categories. The categories and the differences between them were then defined. This was followed by coding the responses in relation to each of the categories. If the response related to more than one category then they were

coded as both and identified in the 'multiple' response category. The responses which demonstrated 'multiple' elements were generally more sophisticated responses. Given the length of responses, a content analysis was not appropriate. Instead, to get a sense of the nature of responses and how these varied by discipline, year and gender, the different categories were totalled in order to explore patterns of similarity or different themes between groups. It was not possible to test for statistical differences on these findings as there were too many distinct categories that could not be collapsed together to meet the assumptions of Chi-square.

Secondly, the MEQ responses were analysed by the students overall 'ethical score', by element and by question. Following Clarkeburn *et al.* (2003) each student received an ethical score. The score was calculated using the following weightings: A=1, Ab=4, B=9, Cb=16, C=25 (Table 3.2). The response to each question was totalled and then divided by the number of questions (10 in total)¹. The resulting figure represented the 'type' of ethical knowledge each student had at that point in time (Type A = 0–8.9, Type B = 9-15.9, Type C = 16-25). The findings were analysed in Excel and SPSS to look for visual and statistical differences. The parametric test of Analysis of Variance (ANOVA) was chosen to test for significance between discipline, year and gender and ethical scores, as it is a more powerful test, and the data approximates to normal distribution for the ethical scores of the MEQ (Mean 16.2; Skewness -0.340; Kurtosis -0.028). Pearson's Chi-Squared test

¹ The data was also analysed using a single scale of 1, 2, 3, 4 & 5 for each answer. As with Clarkeburn *et al.*'s (2003) study the significance of the results was not altered by modifying the values given to the different points on the meta-ethical scale in this way. The data reported uses the weighted scale in order to be comparable with Clarkeburn *et al.* (2003).

was chosen to test for significance between types of meta-ethical development and discipline, year and gender as this was nominal/categorical data.

	Definitely my opinion	More or less what I believe	Neither statement represents my view	More or less what I believe	Definitely my opinion	
Statement Type A	А	Ab	В	Cb	С	Statement Type C

Table 3.2: Meta-ethical Questionnaire response options

3.2.2 Tutor interviews

The fourth research question was addressed through in-depth interviews and the questionnaire material. In-depth semi-structured interviews were employed with two tutors in English and Animal Behaviour. These were tutors of classes who had completed the questionnaires (alongside the data from the 13 interviews with Geography tutors from the previous project). These interviews focused upon the nature of ethical thinking expected by tutors. As student ethical development occurs through the improvement in the ability to think ethically, it is important to understand what tutors expected from their students, in order to interpret the level of development found in the questionnaire.

An in-depth interview is a "conversation with a purpose" (Dexter 1970: 136). Semistructured interviews are considered the most effective way of collecting information on the nature of ethical thinking required in their discipline. This is because interviews enable "opinions, networks of relationships and ideas to be presented and qualified" (Hoggart *et al* 2002: 205). This is important as the level of detail needed to analyse the nature of ethical thinking in different disciplines "is best communicated through detailed examples and rich narratives" (Hoggart *et al* 2002: 205). The advantage of interviews over questionnaires for the discipline tutors was that they are more conversational, and could be varied according to the interests, experiences and views of the interviewees (Valentine 2005). The interviews were semi-structured to encourage discursive dialogues, as Goudge (2003) found "this is the most effective way of exploring underlying attitudes" (p. 96). The interviewees were asked questions about their perceptions of the ethical thinking skills their students require upon graduation. Specifically the interviews related to three themes: understanding (what tutors thought their students should understand about ethics once they have completed a degree in their discipline); relevance (how relevant was or should ethics be to the undergraduate learning within their discipline); and process (how ethics is taught in the discipline). The interviews were recorded on a dictaphone with the permission of the respondents. A full transcription of each interview was produced.

Ethical issue	Interviews
(Source: Hay 2003)	
Consent	• Tutors were invited to participate in an interview via email. Each of these tutors had already assisted the project by providing time in their class to hand out the questionnaire.
	• A Participant Information Sheet (PIS) was emailed to the tutor before the interview and they were given a paper copy at the beginning of the interview (Appendix 2).
	• Each interviewee was asked to complete an informed consent form to state that they understood the information on the PIS, understood that their participation was voluntary, and were willing for the interview to be recorded (Appendix 3).
Confidentiality	 Pseudonyms are used in this write up to protect the anonymity of the interviewees. The full transcripts have not been included in the appendix as the detail of
	the work discussed would reveal the identity of the participant.

Harm	 The PIS made clear that there would be no adverse effects or benefits to the tutors whether they chose to take part in the study or not. Through the course of the interviews, tutors could reveal information that could reflect negatively on them. If this occurred and the information was not directly relevant to the research then it was omitted from the transcription.
Cultural awareness	• The topic of ethics is sensitive with moral attitudes and perspectives being variable between different people and cultures. In the course of the interview it was made clear that I was interested in the participant's view of ethics within the discipline.
Dissemination	 An electronic version of this dissertation will be uploaded to my staff website so that participants may read the findings. An email of the abstract will be sent to the tutors who participated in the interviews with a link to the full electronic final report.

Four in-depth interviews were conducted in addition to the interviews with Geography last year. Two of the interviews were with tutors from English and two from Animal Behaviour and Welfare. The Geography interviews covered all of the core tutors within the department. The interview data was analysed using NVivo, focusing upon the three themes of the interviews: 1) understanding – tutor expectations of their students understanding in relation to ethics; 2) relevance – how relevant tutors perceive ethics to be to their discipline and teaching; 3) process – how ethics was taught in the discipline. This helped to create a rigorous analysis of the data producing a strategic and systematic interpretation of what the tutors said.

The final research question was addressed by discussing the findings from both the interviews and the questionnaire in relation to existing literature on teaching challenging topics. The next chapter discusses student ethical development by discipline year and gender, followed by Chapter 5 which discusses tutor reflections on ethical thinking within different disciplines.

Student ethical development by discipline, year and gender

This chapter focuses upon the analysis of the questionnaire data. It addresses the

first three research questions:

- 1. What are the differences and similarities in the ethical development of students in three academic programmes in the arts, social and pure sciences?
- 2. What is the extent to which there is progression in the ethical development of students in different years across the three academic programmes?
- 3. Is there any difference in the ethical development of students by gender across the three academic programmes?

4.1. Characteristics of respondents

This analysis is based on 335 responses of first, second and third year undergraduates; 258 were from single honours students in English, Geography, Animal Behaviour, Animal Behaviour and Welfare, Biology and Zoo Management, 77 were from students who were taking one of the three subjects combined with another art, social science or science subject respectively (Table 4.1). Thirty-eight of the responses where from students combining with disciplines outside arts, social sciences or sciences, and as such were excluded.

The majority of the students who completed the questionnaire in English were studying single honours English (58 responses, 59%). The same was true for Geography with single honours students representing the majority of responses from the discipline area (52 responses, 60.47%). Within the Animal Behaviour
responses the majority were studying single honours Animal Behaviour and Welfare (86 responses, 56%), the second largest group were studying single honours Animal Behaviour (47 responses, 31%).

Er (Si co an	n glish ngle Honours or mbined with other art)	98	Geography (Single Honours or combined with another social science*)	86	Animal Behaviour (Single Honours or combined with another science)	151
•	English (SH)	58	• Geography (SH)	52	 Animal Behaviour (SH) 	47
•	English and Creative Writing (CH)	15	 Geography and Sociology (CH) 	1	 Animal Behaviour and Welfare (SH) 	86
•	English and Drama (CH)	10	 Geography and Business Studies (CH) 	1	 Animal Behaviour module as part of Biology (SH) 	11
•	English and History (CH)	6	 Geography and Tourism (CH 	1	 Animal Behaviour and Biology (CH) 	3
•	English and English Language (CH)	3	 Geography and Events Management (CH) 	1	 Zoo Management (SH) 	4
•	English and Theology (CH)	3	 IDS and Geography (CH) 	17		
•	English and Fine Art (CH)	3	 NHM and Geography (CH) 	8		
			 NHM and IDS** (CH) 	4		
			 Geography and Spanish (CH) 	1		
					Total responses	335

Table 4.1: Questionnaire responses by programme

*Note: this group only includes students who identified human geography or geography as a whole as their interest. **Although students studying Natural Hazard Management and International Development Studies do not have Geography in their programme title, in many institutions these subject areas would be studied within a Geography programme.

For each subject area, a similar percentage of students responded from the different years (Table 4.2). Around 40% of the responses were from year 1 from each subject. The percentage of responses from years 2 and 3 were both around 30%.

Table 4.2: Responses by year

	Year 1		Year 2		Year 3		Total
	No.	%	No.	%	No.	%	
English	43	44	28	29	27	28	98
Geography	32	37	29	34	25	29	86
Animal Behaviour	61	40	38	25	52	34	151
Overall	136	41	95	28	104	31	335

Overall more women completed the questionnaire (77%) in comparison to men (23%) (Table 4.3). Whereas Geography had the highest proportion of men (42%) both English and Animal Behaviour had much lower proportions (both at 17%). These differences reflect the proportions of men and women studying the disciplines at the case study university (Animal Behaviour: M=18%, F=82%; English: M=24%, F=76%; Geography: M=56%, F=44%). This analysis begins by looking at the quantitative findings from the questionnaire before discussing the responses to the open-ended questions.

Table 4.3:	Responses	by gender
------------	-----------	-----------

	Male		Fen	Total	
	No.	%	No.	%	
English	17	17	81	83	98
Geography	36	42	50	58	86
Animal Behaviour	25	17	126	83	151
Overall	78	23	257	77	335

4.2. Meta-ethical questionnaire

The following discusses the findings from the Clarkeburn *et al.* (2003) meta-ethical questionnaire (MEQ) across all three disciplines, and by year and between genders as a whole. The findings are then analysed by element by discipline, year and gender.

This section of the analysis focuses upon 295 responses. This smaller number of responses has been filtered for two reasons: 1) the MEQ section of the questionnaire was not completed in full (13 respondents); and 2) the response did not pass Clarkeburn *et al.*'s (2003) internal validity check (27 respondents) - where the two statements representing element V (purpose of ethical discussion) were essentially addressing the same point (statement pairs 8 and 9). Responses which had more than one step difference between the two statements were considered invalid. The breakdown of the responses is shown below in Tables 4.4 and 4.5 indicating a similar pattern of response as the whole group.

Table 4.4: MEQ usable responses by year

Subject	Year 1	%	Year 2	%	Year 3	%	Total
English	36	42	25	29	24	28	85
Geography	22	30	28	38	24	32	74
Animal Behaviour	52	38	34	25	50	37	136
Total	110	37	87	30	98	33	295

Table 4.5: M	EQ usable	responses	by	gender
--------------	-----------	-----------	----	--------

Subject	Male	%	Female	%	Total
English	15	18	70	82	85
Geography	30	41	44	60	74
Animal Behaviour	22	16	114	84	136
Total	67	23	228	77	295

The MEQ asked participants to select a position between ten sets of contrasting paired statements (Appendix 1). There were 5 options for each set of paired statements enabling students to position themselves in the middle between the two statements if neither represented their view. According to Clarkeburn *et al.*

(2003) students who indicated that their beliefs were closer to the statements on the right hand side of the questionnaire were demonstrating greater understanding of the complex, uncertain, variable nature of ethical issues, and recognising that such issues are contingent and open to different perspectives. Furthermore the statements to the right suggest greater self-awareness and that the participant takes greater ownership of their decisions. The mean ethical score for all of the MEQ responses was 16.2 with a minimum score of 4.9 and a maximum score of 25. This is comparable with Clarkeburn *et al.*'s (2003) control group² who on scored a mean of 16.9.

4.2.1. Discipline analysis

The mean ethical score for each discipline indicates little difference between the different subject areas (Figure 4.1) with all falling in the narrow range with the highest mean score of 16.6 (English) and the lowest 15.6 (Geography). The Animal Behaviour students received a mean a score of 16.2. Addressing question one this suggests that there is little difference between the current ethical development of students in the three academic programmes.

² The control group in Clarkeburn *et al.*'s (2003) study did not participate in the ethics programme.

Figure 4.1: Mean ethical score by discipline



The findings can be further examined to show the range of ethical scores within the different disciplines (Table 4.6 and Figure 4.2). The broadest range of ethical scores was within Animal Behaviour with a standard deviation from the mean of 3.8 and an inter-quartile range of 5.2. English had the narrowest range with a standard deviation of 3.1 and an interquartile range of 5.1. Although the range for Animal Behaviour is slightly larger, it is only a minor difference.



Figure 4.2: A 'box and whiskers' plot of the ethical scores by discipline

Key: Dark blue = 1^{st} quartile range. Light blue = 3^{rd} quartile range.

Table 4.0. Mean ethical score and range by discipline	Table 4.	6: Mean	ethical	score and	range by	y discipline
---	----------	---------	---------	-----------	----------	--------------

	Mean Ethical Score	Standard Deviation	Median Ethical Score	Q1	Q3	Interquartile Range
English	16.6 (n=87)	3.1	16.6 (n=87)	14.0	19.0	5
Geography	15.6 (n=74)	3.6	15.55 (n=74)	13.5	18.2	4.8
Animal Behaviour	16.2 (n= 136)	3.8	16.2 (n=136)	13.9	19.1	5.2

An ANOVA test to compare the means found that there were no significant difference by discipline (F value 1.629, p < 0.198, df 2). The means from the discipline ethical scores may also be examined in line with Clarkeburn *et al.*'s (2003) use of Perry's (1999) scheme to categorise different students' ethical development based on the following thresholds: Type A = 0-8.9, Type B = 9-15.9, Type C = 16+ (Table 4.7; Figure 4.3).

 Table 4.7: Summary of characteristics of different types of ethical development

Туре А	'Safety in dualism' – sees the world in dualistic terms with clear rights and wrongs, they view diversity as an unwarranted confusion.						
Туре В	'Distress in relativism' – accepted a world of multiplicity and relativism, because they believe that the tutors want them to accept such a world.						
Туре С	'Comfort in commitment' – makes commitments to moral values, taking responsibility for chosen values and how to fulfil them.						

Source: Clarkeburn et al. (2003: 445-447)

Figure 4.3 clearly shows that for English (62.4%) and Animal Behaviour (53.7%) the majority of students can be classified as 'Type C'. This indicates that the majority of these students in all of the three disciplines are recognising the variability and contingent nature of ethical issues. However, a slightly higher proportion of Geography students could be classified as Type B (50.0%) rather than Type C (46.0%). According to Clarkeburn *et al.* (2003) these students continue to struggle with the complexity of multiple perspectives and understandings of different issues. Perry (1999) suggests that most students who have reached the stage of higher

education are making the transition from Type B to Type C (cited Clarkeburn *et al.* 2003). Type A students are "expected to be rare in the higher education population" (Clarkeburn *et al.* 2003: 447) given that a by the time a student reaches university education they are likely to have been taught to think critically to some degree about the subject material of the different disciplines. These findings support this with less than 5% of Geography and Animal Behaviour students, and no English students being classified as Type A. This suggests that these students are applying these critical thought processes in considering how they think about and manage ethical issues. A Pearson Chi-Square test indicates that there is no significant difference between the three different disciplines and the types of meta-ethical development the students have demonstrated (Chi-Square value 4.322, p < 0.115, df 2). The overall MEQ analysis in relation to research question one demonstrates that there are no significant differences between the three differences between the three disciplines.



Figure 4.3: Participant ethical scores by different types of ethical development and discipline

4.2.2. Year group analysis

As argued in the literature review, the extent to which students might progress in their ethical development as they go through their studies is questionable (Dellaportas 2002; Ludlum & Mascaloinov 2004; Perry 1999). This research has tested the assumption that students will be at different stages of ethical development at different points in their university studies using various different methods of analysis. These different methods of analysis have indicated some variation in the answer to research question two.

An analysis of the means for all three disciplines at each of the different years of the undergraduate programmes indicates a lack of difference between the years (Table 4.8 and Figure 4.4). Year 3 has the highest mean at 16.8; however this is only higher by 1.4 than the lowest score at year 2. Furthermore the overall findings lack a clear progression from first year to the third year of the undergraduate programme. Focusing upon question two, this suggests that there is relatively little, if any, progression in ethical development across the three years of a programme. This implies that development cannot be directly linked to how far along they are in their programme of study. However, this research was with different cohorts at the same point in time, the differences between these year groups therefore relates to cohort differences rather than a lack of progression for individuals. A longitudinal study with the same cohort would be needed to demonstrate individual development.

Figure 4.4: Overall year group differences



Table 4.8: Mean ethical score by year

	Year 1	Year 2	Year 3	Overall
English	16.7	16.4	16.9	16.6
Geography	16.1	15.3	15.6	15.6
Animal Behaviour	16.2	14.8	17.2	16.2
Overall	16.4	15.4	16.8	16.2

The findings for each year group can be examined further to analyse the range of scores within each group (Table 4.9 and Figure 4.5). This illustrates that the median ethical score increases by year. The standard deviation from the mean is similar for all three years (max 3.6, minimum 3.5) as is the interquartile range. Although they might suggest some possibility of progression, this is perhaps misleading, particularly given that the highest ethical score of 25 was for a student in the first year of their degree, and the lowest of 4.9 was for someone in their final year.

	Mean Ethical Score	Standard Deviation	Median Ethical Score	Q1	Q3	Interquartile Range
Year 1	16.4 (n=110)	3.5	16.3 (n=110)	14.0	18.9	4.9
Year 2	15.4 (n=87)	3.6	15.3 (n=87)	13.3	18.2	4.9
Year 3	16.8 (n=98)	3.6	17.1 (n=98)	14.9	19.3	4.4

Table 4.9: Mean ethical	score and	range k	y year
-------------------------	-----------	---------	--------

Figure 4.5: A 'box and whiskers' plot of the ethical scores by year



Key: Dark blue = 1^{st} quartile range. Light blue = 3^{rd} quartile range.

An ANOVA test to compare the means found that there was a significant difference by year (F value 3.612, p < 0.028, df 2). This demonstrates the differences between year 2 and years 1 and 3, suggesting that either there is no clear progression across the degree programme or that there are differences between the cohorts. This can be further analysed to investigate students' levels of ethical development at different points in the discipline programmes. This indicates some differences between the disciplines. For English and Animal Behaviour the year 3 students have developed greater awareness of the complexity of ethical issues. However all of the disciplines dip slightly in year 2. The largest difference is between Animal Behaviour in years 2 (14.8) and 3 (17.2) (Table 4.8, Figure 4.6). This may relate to the nature of the programme in Animal Behaviour. Of the response from third year Animal Behaviour students, 49 of the 52 completed the questionnaire in the first session of a module entitled: 'Ethics, Welfare and Applied Animal Behaviour'. This is an optional third year module, although the majority (91%) of students on the Animal Behaviour programme elect to take it. The clear emphasis upon ethics within the module may mean those students who elected to take the module are more interested in the ethical issues.





When the ethical scores by year are divided into the different types of meta-ethical development there also appears to be some progression towards greater ethical development by the third year of the degree (Figure 4.7). The majority of students in all three year groups were classified as Type C, the second highest percentage as Type B, and a small minority categorised as Type A. Year 3 had the highest percentage of students categorised as Type C (62%). Year 2 had a slightly lower proportion of students as Type C (49%) than Year 1 (57%). However, year 2 had more Type B students (47%) than any other year. A Pearson Chi-Square test indicates that there is a significant difference between the three different years and the types of meta-ethical development the students have demonstrated (Chi-Square value 7.537, p < 0.023, df 2). The overall MEQ analysis in relation to

research question two demonstrates that there are significant differences between the three year groups. However, these findings suggest differences between year 2 and years 1 and 3, this is unlikely to be direct progression between the years, but it may reflect cohort differences.



Figure 4.7: Participant ethical scores by different types of ethical development and year

4.2.3. Gender analysis

As the literature review demonstrated, although the majority of research has found that with the tools being utilised in similar studies, females often appear more developed in their ethical decision making than their male counterparts (Miori *et al.* 2011), there are exceptions (Kelger 2011; Phau & Kea 2005). This research found that women consistently received a higher ethical score (16.4) than the male participants (15.7) (Table 4.10, Figure 4.8). However, the difference is small. Using various different methods of analysis, this research has found some variation in the answer to research question three.

Table 4.10: Mean ethical score by gender

	Male	Female	Overall
English	16.2	16.8	16.6
Geography	15.4	15.8	15.6
Animal Behaviour	15.8	16.3	16.2
Overall	15.7	16.4	16.2

Figure 4.8: Mean ethical score by gender



Figure 4.9 and Table 4.11 illustrates how the standard deviation (4.2) and interquartile range (5.1) are slightly higher for the male respondents than females (standard deviation 3.4, interquartile range 5.0), with the median score being slightly higher at 16.3 for females (males 16.0).

Table 4.11: Mean ethical	score and range	by gender
--------------------------	-----------------	-----------

	Mean Ethical Score	Standard Deviation	Median Ethical Score	Q1	Q3	Interquartile Range
Male	15.7 (n=67)	4.2	16.0 (n=67)	13.5	18.6	5.1
Female	16.4 (n=228)	3.4	16.3 (n=228)	13.9	18.9	5.0

Figure 4.9: A 'box and whiskers' plot of the ethical scores by gender



Key: Dark blue = 1^{st} quartile range. Light blue = 3^{rd} quartile range.

An ANOVA test to compare the means found that there was no significant difference by gender (F value 1.910, p < 0.168, df 2). When the means by gender are analysed by discipline the female respondents continue to receive the highest mean scores. However, the differences for Geography are only 0.5 (Table 4.10, Figure 4.10).





Overall, the majority of the responses from both genders were categorised as 'Type C' (male 51%, females 55%) (Figure 4.11). More of the female responses (43%) were classified as 'Type B' than male responses (42%), alongside a lower percentage classified as 'Type A' (1%) than male responses (8%). A Pearson Chi-Square test indicates that the difference between gender and the types of meta-ethical development is significant (Chi-Square value 7.442, p < 0.024, df2). The overall MEQ analysis in relation to research question three demonstrates that there is a significant difference between the genders with females being more ethically developed (using this tool) than their male counterparts. This supports the majority of studies that have found females to be more ethical than males (Miori *et al.* 2011).



Figure 4.11: Participant ethical scores by different types of ethical development and gender

4.3 Meta-ethical questionnaire elements

Clarkeburn *et al.* (2003) based the questions in the Meta-ethical questionnaire on five elements which assessed different types of development (Figure 4.12 & Table 4.12). The questionnaire has been analysed by individual question and by each element. The findings presented here focus on the elements as these offer a focus on key themes and are enable comparison with Clarkeburn *et al.*'s study. Elements I, II and V have similar MEQ means ranging between 17.6 and 18.4, whereas Element III and IV are lower (12.7 and 14.1 respectively) (see Appendices 4-8 for the findings by discipline, year and gender).





A Chi-square analysis of the student scores between each of the different elements indicates that nine out of ten of the scores were significantly different (Table 4.12) (the exception being the difference between Element III and Element IV). This relates to the fact that the different elements were assessing different types of development. 'Element III: Nature of multipliticy' (12.75) was a lower mean than the mean of the control group in Clarkeburn *et al.* (2003) (18.03). Whereas 'Element II: Role of authority' received a higher mean (17.69) than Clarkeburn *et*

al.'s (2003) control group who received 15.14³. These findings suggest that, overall students were slightly more comfortable with the notion that authority figures did not have the 'right' answer than the students in Clarkeburn *et al.*'s study. However, the students in this study were less comfortable with notion of multiplicity. These differences may relate to the way in which students are taught at the two different institutions. The smaller class sizes at the case study university may lead to greater discussion with the tutor and recognition that the tutor does not provide all the answers. The relative lack of comfort with multiplicity may relate to the age of the majority of the participants (85% of all of the participants were 21 and under), and their lack of life experience.

Element	Mean		Element	Element	Element	Element	Element
			I	II	111	IV	V
Flomont & Course and		P=		0.000	0.000	0.000	0.000
type of moral answers	17.6	Chi-Square		51.537	43.912	28.007	47.670
type of moral answers		df		4	4	2	4
Element II: Polo of		P=			0.016	0.006	0.000
authority	17.7	Chi-Square			12.174	14.408	70.805
authority		df			4	4	4
Flomont III, Naturo of		P=				0.053	0.028
multiplicity	12.8	Chi-Square				5.892	10.836
muniplicity		df				2	4
Element IV: Personal		P=					0.010
responsibility and	14.2	Chi-Square					13.181
relationship with multiplicity	14.2	df					4
		P=					
Element V: Purpose of	18.4	Chi-Square					
moral discussions		df					

Table 4.12: Statistical differences between elements

However, when each element was analysed by subject, year and gender there were only two significant differences out of fifteen tested (Table 4.13). For Element III there was a significant difference by gender. Appendix 6 illustrates how although

³ The remaining means were within ±1.25 of the means found in Clarkeburn *et al.*'s (2003) study.

the mean score for men and women in the different disciplines were similar, the distribution of scores indicate the female scores peak much higher (47.3%) with more of them believing that there are few 'absolute right answers' than males (33.5%). This difference supports earlier research that found females to have greater ethical understanding than males (Ludlum & Mascaloinov 2004). Conner (1999) argues that women can focus on more than one problem at a time, relative to men, and that they generally prefer to solve problems through multiple activities. This may relate to the greater comfort the women in this study had with the nature of multiplicity.

Element	Mean		Subject	Year	Gender
Flowent I. Course and tune of morel		P=	0.154	0.185	0.261
element I: Source and type of moral	17.6	Chi-Square	6.676	6.199	4.001
allsweis		df	4	4	3
		P=	0.431	0.720	0.051
Element II: Role of authority	17.7	Chi-Square	5.931	3.678	7.773
		df	6	6	3
	12.8	P=	0.789	0.098	0.045
Element III: Nature of multiplicity		Chi-Square	4.703	13.437	9.754
		df	8	8	4
Flowert N/, Deversel versesibility and		P=	0.086	0.697	0.550
Element IV: Personal responsibility and	14.2	Chi-Square	4.913	2.210	1.197
relationship with multiplicity		df	2	4	2
		P=	0.595	0.004	0.315
Element V: Purpose of moral discussions	18.3	Chi-Square	2.784	15.274	2.308
		df	4	4	2

Table 4.13: Statistical analysis of elements by subject, year and gender

For Element V there was a significant difference by year. Appendix 8 illustrates that for all disciplines 59.5% of year 3 students definitively agreed that moral discussion was valuable and enjoyable even if the group did not agree on one right answer, in comparison to 45.6% and 40.0% of students in years 1 and 2 respectively. This significant difference may relate to their greater comfort in discussing issues generally within their studies as they progress in their degrees.

4.4. Open-ended questions

The ethical score offers a quantitative measure for interpreting differences between disciplines, years and gender. The open-ended question section complements this analysis by providing a sense of the way in which students understood ethics in their own words. This section focuses on explaining the main differences.

4.4.1. What is ethics?

When asked to define what ethics was, the responses fell into eight main categories (excluding miscellaneous responses or no answers). Student responses to this question varied from simplistic answers which defined ethics as what was 'right' or 'wrong' to more sophisticated responses which highlighted the contextual nature of an issue, and how the background of the individual influenced how people perceived what was ethical in a given situation (Table 4.14).

Category and definition	Example from each discipline
Morals	I thought that 'ethics' was something moral, something ethically
Some mention of morals/morality.	correct.
These responses did not often take a	In reference to morals and morality of actions.
position as to what morality meant,	It's like another word for moral if that makes sense? Like saying
but connected the notion of being	something isn't ethical means it's not morally correct.
'moral' to ethics.	

Table 4.14: What is ethics? Categories and examples

Right or wrong	Ethics is a code of behaviour based on a person believes may be right
Where ethics was interpreted as	or wrong.
knowing whether something was right	
or wrong, or where people follow	The difference between right and wrong. What is right and what is
guidelines which stated whether	wrong.
things were right or wrong. These	
responses did not recognise the	
contingent and contextual nature of	The right or wrong way to do something.
now what is right and wrong might	
Valy.	t shink shine and earthin unumittee mules shot weards south for
Benaviour/values	I think ethics are certain unwritten rules that people create for
othics influences their behaviour or	themselves or within an organisation and try to aunere to.
their values Responses were	Beliefs and thoughts of an individual that influences their behaviour.
categorised as behaviour when they	
recognised how ethics impacted on	What you as an individual decide or perceive is right and wrong based
how people might behave, providing	on morals and upbringing; varied different beliefs personal
some recognition of the complexity or	perspective on different issues.
variation between people.	
Animal rights	Whether something is ethical e.g. animal testing etc.
Relating to the rights of animals or	-
non-human organisms.	Deeple's opinions on selected subjects being studied. Such as animal
	testing etc.
Human rights	I think of how people treat each other when I hear the word 'ethics' I
Discussion of people's rights (e.g. the	think of equality, human rights and being fair to one another.
right to choose, the right to a better	That everyone is treated the same. Making sure everyone's needs are
life etc.)	catered for
	People's rights, freedom of speech, the right to choose.
Culture	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within
Culture Some mention of the nature of culture	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards.
Culture Some mention of the nature of culture in their response. Recognising that	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race,
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied. Multiple	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain. Ethics means to me, the subject of morals and whether a certain
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied. Multiple Numerous responses highlighted	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain. Ethics means to me, the subject of morals and whether a certain outcome of circumstances is the right thing to do not only in relation
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied. Multiple Numerous responses highlighted more than one of the above. These	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain. Ethics means to me, the subject of morals and whether a certain outcome of circumstances is the right thing to do not only in relation to the people involved but others' values as well.
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied. Multiple Numerous responses highlighted more than one of the above. These were generally more complex and	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain. Ethics means to me, the subject of morals and whether a certain outcome of circumstances is the right thing to do not only in relation to the people involved but others' values as well. Making the right decision dependant on your personal views. Taking
Culture Some mention of the nature of culture in their response. Recognising that ethics relates to the cultural background of an individual. Research Mention of research directly, or implied from way they have described ethics. Subject specific A response which is specific to the subject being studied. Multiple Numerous responses highlighted more than one of the above. These were generally more complex and sophisticated responses.	People's rights, freedom of speech, the right to choose. A set of unspoken rules to which we adhere in order to remain within social standards. Ethics is related to the background of a person: religion, age, race, gender. When something may be a sensitive subject. We don't want to harm or go against religion etc. Issues and/or debates concerning the treatment and welfare of participants in studies or under observation. Being polite when interviewing/questioning. Keeping an open mind. Not imposing your own opinions on something. The level at which you can carry out 'research' without interfering with privacy. - Issues and subjects which explore the dilemmas that are raised i.e. carbon foot printing etc. The equal treatment of all living things to the best of the ability to survive e.g. cows would probably not exist if they were not used by humans but they should be awarded respect whilst alive and killed humanely with no stress or pain. Ethics means to me, the subject of morals and whether a certain outcome of circumstances is the right thing to do not only in relation to the people involved but others' values as well. Making the right decision dependant on your personal views. Taking other people into account. Being morally correct. Being and feeling

	Personal beliefs in wellbeing of humans/animals. Often personal opinion what is/is not acceptable in behaviour etc. based on laws, culture etc.
Miscellaneous	Races and nationality, also whether something is ethically fair
These responses did not relate to the	meaning general rights of people.
concept of ethics or indicated clear	Ethnicity: racial characteristics. Heritage/background. Ethics: to treat
misunderstandings of the term.	everyone equally. To be considerate of others views.
	I generally think of other races than British white, or something that is
	different.

The two most common categories for all three disciplines, years and genders were 'morals' and 'right or wrong' (Figure 4.13). The percentage of English students who referred to morals in some way (48%) was higher than that of Geography (23%) or Animal Behaviour (31%). This difference may relate to experience of using the term 'moral', with English students being more familiar with the term due to its use in their studies, as opposed to discussing ethical reasoning or decision making.

In a large number of these responses (English 18%, Geography 19%, and Animal Behaviour 21%) students had combined the two, for example "Ethical - what is the right thing to do? The moral path." Yet the relationship between 'moral' and 'right and wrong' in the majority of responses illustrated how students had a superficial understanding of the connection between the notions of morality and ethics. As argued earlier morals relate to the personal values of an individual, rather than recognition of the broader social context. The link therefore between right and wrong suggests that students are recognising only their individual values and not that these might vary between people (this would be the equivalent to the development of Element III in the MEQ).

More sophisticated responses recognised that ethics was broader than the individual discussing behaviour or values, or related the ideas to animal and human rights. These were close to demonstrating how an individual's morals were played out and hence how they impacted upon a broader notion of ethics. English (11%) had the highest percentage of responses which related to behaviour or values (Geography 8%, Animal Behaviour 8%). This was differentiated by gender with women (10%) noting behaviour and values more than men (5%). As might be expected a higher percentage of responses from Animal Behaviour students (10%) related their response to animal rights (English 4%, Geography 0%), whereas a slightly higher proportion of Geography students (9%) related their response to human rights (English 5%, Animal Behaviour 5%). Geographers (11%) also showed greater awareness of ethics within research (English 2%, Animal Behaviour 2%). At the time of the questionnaire the second and third year geography students had recently undertaken work in which they had to identify potential ethical issues in their research and so were familiar with the term ethics being used in connection with research. It was also notable that year 3 Animal Behaviour students had longer, more in-depth explanations as to what ethics was, perhaps reflecting the module they were studying. In summary this analysis demonstrates that overall there are few differences between disciplines, years or genders as to the definition of ethics. The majority of students had a basic understanding of what ethics was. However this was generally focused around word association about the term ethics as opposed to a recognition of the complexity of what 'ethics' encompasses.

50



Figure 4.13: What is ethics? Percentage of responses in each category by discipline, year and gender

4.4.2. Example of ethical issue previously studied

This question asked students to identify a specific example of where they had studied ethics in their previous education. The categories for the examples related to who was involved in the ethical issue, for example people (human rights), the environment (environmental concerns) or animals (animal rights); alongside broader areas relating to politics, culture and general research (Table 4.15).

Category and definition Example from each discipline **Human rights** Euthanasia, do we have any right to decide how and when to die. An example of studying something to Exploitation of children in the clothing business and education. Is it do with a person's rights (e.g. the right right they miss school to work and provide us with cheap clothing or to choose, the right to a better life). are we providing them with a future. Abortion - good/bad, Euthanasia - good/bad. **Environmental concerns** Recycling An example of studying something Ethical issue of study carbon footprint of field trips, fairness in modern which relates to the environment. society. Greenhouse gas emissions - use of cars etc. at our leisure. Existence of zoos, captive animals. **Animal rights** Drug testing on animals. An example of studying something Ethical treatment of animals. relating to the rights of animals or non-human organisms. Wild caught animals being sold within the pet trade. Is it right to In GCSE I studied the rise of Hitler and WWII in history. Political An example of studying something China's one child policy. relating to specific political events, or Abortion debates, euthanasia, capital punishment broader political discussions. Culture Which religion is right to you, what you believe (your opinion) in and An example of studying something what do you believe in. that related to culture or religion in Religion - comparing various aspects of the church, their views and terms of ethics. what they stand for. Research When creating a questionnaire I made it anonymous so it was fair to An example of considering ethics my fellow students. directly in relation to research, or Confidentiality while doing interviews in Slapton, Devon. implied from way they have described Is right to breed animals solely for the purpose of using them in ethics. Subject specific Drama whether everything is performance i.e. wanking on stage, An example of ethics which was burning one's self, removing scroll from lady parts. specific to the subject being studied. How NGOs and charities have helped out LEDCs after natural disasters e.g. tsunami.

Table 4.15: Example of ethical issue previously studied: Categories and examples

	The ethics of genetically altering an animal e.g. is it fair to have a featherless chicken to survive in hot countries to provide people with food? Chicken wouldn't be able to exhibit natural behaviour (i.e. preening) but is the sacrifice allowable to save human lives?
Multiple	In psychology, particularly whether people or animals were treat in an
An example which touches upon	ethical way while a study had been carried out, for example a study on
multiple elements of the other	operant conditioning by Pavlov on dogs, we had to argue whether the
categories.	treatment of dogs were ethical.
	Issue of buying flowers from supermarkets as the long lasting flowers have chemical sprays used and can cause infertility problems and respiratory problems to the employees in developing countries.
	Should animal testing be allowed, death penalty, hunting ban.
Miscellaneous	Gender.
An example which does not fit the	Theory of knowledge.
other categories and the relevance or	
connection to ethics is unclear.	Age in psychological studies.

When it came to examples from previous studies there was greater variety between the different disciplines (Figure 4.14). The largest number of examples came from Animal Behaviour students (48%) who identified an example of a previous study of ethics related to animal rights (English 8%, Geography 3%). Given the subject area, this is not a surprising finding. However, over the three years there is an interesting parallel between a decrease in the examples of human rights identified (Year 1: 36%, Year 2: 32%, Year 3: 12%) as examples of animal rights increase (Year 1: 20%, Year 2: 24%, Year 3: 37%). Predominantly this relates to the Animal Behaviour students and their greater recognition of animal rights as ethical issues at different stages of their degrees. As in their definitions of ethics, a higher proportion of Geography students (22%) provided examples of ethics in relation to research (English 13%, Animal Behaviour 12%). Overall there are few differences between disciplines, years and genders in terms of the ethical issues they had discussed prior to University.



Figure 4.14: Example of ethical issue previously studied: Percentage of responses in each category by discipline, year and gender

4.4.3. Example of ethical decision

The final open question of the questionnaire asked students to identify an example of an ethical decision that they had made in the past. The examples categorised in Table 4.16 were often short statements without much detail about the reasons why a decision or choice involved the consideration of ethics, or how it was dealt with in an ethical manner. Often the examples illustrated general decisions or choices, rather than necessarily *ethical* decisions or choices. In hindsight greater explanation as to what was meant by an 'ethical' choice might have helped here. However, it may have been that students either did not feel that the choices they had made were underlain by ethics, thereby indicating a lack of understanding of the extent of ethical issues within society, or they were concerned about being judged by the choices they had made.

Category and definition	Example from each discipline
Research	I decided to fill in this questionnaire.
An example which involved some form	Keeping the names of people I have interviewed confidential.
of research.	Use of invertebrates during lab practicals -put through stressful
	conditions and disposed of but not killed - should they have been used and should they have been humanely killed?
Academic	I could have looked at the answers to a test in school before taking it,
An example which involved academic	but chose not to as I have a strong guilty conscience.
issues.	Human or physical? [the choice between which type of geography to take]
	To not cheat on exams or tests.
Personal	A guy called me a name so I was undecided whether to punch him.
personal issues.	Sex and education.
	Not becoming vegetarian.
Social	Whether to grass my friends out on school property. I chose to notify
An example which involved dealing	a teacher.
with other people.	A very distinct and problematic lack of integration at high school between Pakistani Muslims and the rest of the student population.
	Standing up for people being bullied.

Table 4.16: Example of ethical decision: Categories and examples

Employment	Not to work, and be supported by my boyfriend and sister felt
An example which involved some form	unethical, but needed consideration towards negative distractions.
of employment, whether paid,	working at Oxfam - asking people for money to donate to charity. I
voluntary, or as part of their studies.	and to the right place
	On work based learning
- · · ·	
Environment	-
considering the environment.	Buying recyclable products or products that have been recycled
	(paper).
	Ride my bike or make instead of driving a car.
Animal rights	Have used make up that is not tested on animals.
An example which involved considering the animal rights.	Vegetarianism: I chose not to eat meat because of ethical issues and environmental issues.
	Not buying eggs on campus as they are not 'free range'.
Human rights	Profile/prochoice for abortion and how many weeks.
An example which involved considering the human rights.	Should I choose fair trade coffee/product when the opposite choice appears cheaper?
	The rights go to university and learn without prejudice and restrictions, financially or by health issues.
Culture	-
considering culture within an issue.	Respecting different religious backgrounds and different opinions and controversial issues such as abortion, stem cells, etc.
Multiple	Do I lend my notes to someone who couldn't be bothered to make
An example which touches upon more	their own in seminars?
than one of the categories above.	Someone needed help with work but they started copying mine - had
	to choose whether to let them copy or not.
	Not keeping an animal (rabbit) in a university house. I thought of the
	ethics in this choice as it would not have the best life.
Miscellaneous	·
the other categories and which lacks a	Which university to attend in relation to statistics and environment.
clear explanation as to how it relates to ethics.	Moving away from family.

For all of the disciplines, years and both genders, the majority of students did not provide a response. More men (51%) chose not to answer than women (41%) with 70% of English, 44% of Geography and 33% of Animal Behaviour students not providing an answer. This is an interesting difference considering the marginally higher score the English students received for the MEQ element of the questionnaire, yet such a high percentage were unable (or unwilling) to provide an example of an ethical decision they had made in the past. There was also a clear decrease in the number of 'no answer' responses as the years increase (Year 1: 58%, Year 2: 40%, Year 3: 30%). This suggests that students further into their degrees were more confident in answering the question or had a greater understanding of what an answer entailed.

Animal rights once more dominated the responses for the Animal Behaviour students who had provided answers with 28% identifying a decision which related to animal rights (English 6%, Geography 1%). This was dominated by the Year 3 and female students. The third highest percentage in relation to ethical decisions related to research. This was highest for Geography (17%), closely followed by Animal Behaviour (16%), however English was much lower at 2%. This might be a reflection of the way in which research is understood in the arts in comparison to the social and pure sciences. Overall this has illustrated few differences between the disciplines or genders in terms of ethic decision making examples, however the reduction in the number of no answers by year, at least suggests that there is some progression in confidence to answer the question.



Figure 4.15: Example of ethical decision: Percentage of responses in each category by discipline, year and gender

4.5 Summary

This chapter has focused on the first three research questions (Table 4.17). Overall the findings suggest few differences between disciplines, with none of the metaethical development findings being statistically significant. However, some differences were noted in the open-ended questions with regard to the English students' definitions of ethics, relating to morality rather than right and wrong as was the case for the majority of Animal Behaviour and Geography students. There are some differences by year; however some of these differences may illustrate differences between cohorts rather than progression. As with previous research women have been shown to be more ethical than their male counterparts overall, alongside being more comfortable with the nature of multiplicity.

Research question	Key findings
What are the differences and similarities in the ethical development of students in three academic programmes in the arts, social and pure sciences?	Similar mean ethical scores for all three disciplines
	 No significant difference between disciplines and types of meta-ethical development
	 English students identified morals in their definition of ethics more than other disciplines
	 Animal Behaviour students identified animal rights as an example of ethics within their previous studies more than any other discipline
What is the extent to which there is progression in the ethical development of students in different years across the three academic programmes?	 Similar mean ethical scores for all three years
	 A significant difference between years and types of meta- ethical development
	 A significant difference by year for Element V the purpose of ethical discussion
	 The percentage of 'no answers' to providing an example of an ethical choice decreased by year
Is there any difference in the ethical development of students by gender across the three academic programmes?	 Similar mean ethical scores for both genders
	 A significant difference between gender and types of meta- ethical development
	 A significant difference by gender for Element III nature of multiplicity

Table 4.17: Research questions key findings

Tutor reflections on the nature of ethical thinking within the disciplines

This chapter discusses the tutor interview material and how it aligns with the findings on the development of their students' ethical thinking. Hence it addresses the fourth research question: 'What are the differences and similarities in the nature of the ethical thinking expected by tutors between the three academic programmes and to what extent does this align with the students' ethical development?' In Chapter 2 'meta-ethical development', as assessed by the questionnaire, was defined as how students construct ethical reality, how they interpret the nature of ethical properties, attitudes and judgements. This relates to the concept of 'ethical thinking' whereby how students construct ethical reality influences their ability to think ethically. This chapter explores tutor expectations of ethical thinking, comparing these with the way in which the students constructed relevant elements of ethical reality in the questionnaire findings.

Ethics was considered relevant to all of the disciplines, albeit to different extents. The tutors' interpretation of ethics generally related to ideas of 'ethical sensitivity' (Clarkeburn *et al.* 2002), in which students perceive the ethical implications of a situation. As noted in Chapter 2 whilst ethics is a part of the Bioscience and Geography benchmark statements it is absent from the English statement. However, interviewee English 1 argued that: "Literature is the record of human experience, and therefore ethics ought to be enshrined, as at least something that someone ought to be aware of. All the time we are teaching empathy. Literature is about getting inside somebody else's mind and understanding their, perspective, and to do that without an ethical benchmark seems weird."

Interviewee 'English 1' argues convincingly why ethics is fundamental to the discipline. Yet ethical awareness is not mentioned within the subject benchmark statement and is absent from the programme specification suggesting that the relevance of ethics to English is marginal. The difference between the relative importance of ethical sensitivity here appears to relate to the extent of explicit discussions about ethics within the subjects. Whereas in the Geography and Biosciences benchmark statements (QAA 2007a; 2007b), ethics is highlighted regularly in relation to predominantly empirical research, ethical standards and professional codes of conduct. These are the points at which ethics is discussed explicitly. In English clear ethical guidelines are not needed to support empirical research in the discipline. As undergraduate English students are not normally involved in empirical research involving human or animal subjects, there may be a perception that an explicit discussion on research ethics is unnecessary. Yet this is a rather narrow interpretation of research ethics.

Despite the differences in the subject benchmark statements all three disciplines addressed, or claimed to address, ethical issues within their subject areas. However, they were rarely explicit about doing so, with the exception of the need for ethical approval for dissertations which were working with people or animals. An implicit approach to teaching about ethics limits opportunities to engage in

61

'moral reasoning' (Clarkeburn *et al.* 2002), in which students use practical problem solving strategies to make a judgement about a course of action. In English it was explained how: "it's more important in terms of an issue that emerges from the literature that we're studying rather than as stand-alone issue itself" (English 2). The other two disciplines also indicated how ethics was an issue which emerged from the subject content rather than discussed separately. When shown the findings from the questionnaire and the similarities between English and the other two disciplines, one of the English tutors questioned: "maybe we are teaching ethics by stealth?! It's just utterly embedded" (English 1). Animal Behaviour and Welfare commented in a similar manner:

"it's all very important, and relevant, but I think we would all probably say it's a bit like key skills, it's embedded, you know it's in there somewhere" (Animal Behaviour 2).

Although there are taught prescribed modes of behaviour required when looking after animals, ethics and helping students to think ethically is predominantly embedded into teaching and learning rather than made explicit. The geography tutors agreed with this and raised the question as to whether it is appropriate to approach teaching ethics in a different way:

"We do discuss ethical issues, it's just that it is not packaged as ethical issues, and do we really want to go down the route of drawing it and saying its ethical issues?" (Geography 5) This is an important and interesting question, especially given the findings from the questionnaire and a lack of difference between the three subject areas. Highlighting issues as 'ethical' may discourage student enthusiasm for a topic, yet by doing so it is possible to engage in moral reasoning about the issue. By not explicitly discussing the ethical implications of an issue, the opportunity to enhance students' sensitivity to potential ethical issues is also reduced. The question as to whether this is appropriate for each of the different disciplines requires a discussion of the types of ethical thinking different subjects wish for their students to have by the time they graduate.

As argued in Chapter 2, higher education has an important role to play in developing students with the ability to think ethically (Hay & Foley 1998; Smith 1995) recognising moral reasoning to be part of professional responsibility (Boni & Lazano 2007; Nicholson et al. 2009; Robinson 2005). Although students arrive at university with varying degrees of ethical development, their higher education degree should allow them to leave with an enhanced level of ethical sensitivity and moral reasoning, as this will help students to manage the supercomplexity they face when they graduate (Barnett 2000). The case study university establishes this as an institutional graduate attribute, yet the findings from the MEQ suggest there is work to be done to support students' ethical development. This raises the question as to whether ethical reasoning should be a part of disciplinary programmes. None of the programmes explicitly identify 'ethical reasoning' as either a programme outcome or a graduate characteristic. Rather the nature of the ethical thinking tutors discussed as being an expected outcome from their programmes related to more generic abilities which could be applied to ethical issues.

The tutors highlighted some similarities in the ethical thinking they wanted their students to develop. One such similarity was the ability of their students to reflect

63

upon issues from different perspectives. Animal Behaviour and English both highlighted the skill of empathy:

"one would hope that they acquire empathy for the welfare of animals, it is embedded in their course - the concept of welfare" (Animal Behaviour 2)

"[it's] a sort of process during the course of that module of watching them change and mature and develop skills of empathy and understanding that perhaps they haven't had previously" (English 1)

For Geography, it was argued that students should learn to reflect upon the rights of others in research:

"I would expect some of the physical geographers to have a grasp or an awareness of ethics such as when it came to trespassing when doing field work" (Geography 3).

All three disciplines required students to consider the perspectives of others. This relates to developing students' ethical sensitivity whereby they can perceive the ethical implications of a situation. However, the questionnaire found that Element III of the MEQ, addressing the nature of multiplicity, had the lowest mean score (12.75) of all the elements. This suggests that ethics as 'empathy' was the weakest element across all three disciplines. This might be addressed by making ethics within teaching more explicit, but whether this is a desirable teaching strategy is more debatable. One way round this might be to make ethical issues more explicit towards the end of studying a topic. As one of the geography tutors suggested, ethical thinking could be identified as a specific skill for students, and through the completion of a skills audit with students at the beginning and end of the year, ethical thinking could be explicitly identified in their past studies (Geography 5). These findings indicate that more needs to be done to support students in
recognising how different people perceive issues in different ways. This is discussed further in Chapter 6.

The disciplines were also similar in their concern for the ability of students to recognise the complexity of ethical issues independently. Animal Behaviour 1 reflects how:

"one of the funny things I've noticed, is that they - the students that we get - obviously love animals and they love going to the zoo, and nobody ever seems to question the ethics of the zoo which I always find, almost worrying really. Because I worry about zoos but they don't seem to, it's within our culture, and they accept zoos where they wouldn't accept something like vivisection" (Animal Behaviour 1).

In discussing the support given to help students with their dissertation, one tutor commented: "We are almost spoon feeding them, saying this is what you need to consider" when applying for ethical approval (Geography 5). This referred to a list of 5 key areas (informed consent, anonymity, confidentiality, cultural awareness, and dissemination) which students have to consider when applying for ethical approval to conduct their dissertation projects. Using the example of lan McEwan's *Atonement*, English 2 noted:

"I would hope that by the end they would be steering towards a more complex idea of morality and not opting for a simplistic: 'I want somebody to teach me the right answer to this approach'."

As already mentioned, the mean for Element III which addressed the nature of multiplicity by asking students to position themselves in relation to the extent to which moral questions had absolute right answers, was the lowest of all the elements. This suggests that there is still some work to be done to help students to

recognise complexity in moral issues. In terms of thinking through issues independently, Element IV highlights personal responsibility, which had the second lowest mean MEQ score (14.15) suggesting that students are still developing in their abilities to think through issues themselves.

The disciplines, however, were distinct in the way in which their ethical thinking was assessed. English and Geography both aim to support students to become selfaware and reflective, rather than assessing them on specific disciplinary behaviour or knowledge:

"There would be no requirement really for them to be assessed as moral individuals" (English 2)

"It's just more of making them self-aware of some of the choices they make" (Geography 6) In contrast, Animal Behaviour students are assessed directly on their disciplinary knowledge: "They are assessed on their ability to handle an animal in the right way. Which I would say has an ethical element in it" (Animal Behaviour 1). As Animal Behaviour 1 points out, the way in which an animal is treated is underlain by ethical perspectives. However, in contrast to the other two disciplines, there are particular rules and laws in place, for example the 'Five Freedoms' that animals are legally entitled to. Students are assessed on their knowledge and understanding of these laws (Animal Welfare Education 2012). This focus on more prescribed ethical thinking and practice did not however draw out differences between disciplines in the students' ethical scores from the questionnaire. The nature of the ethical thinking skills highlighted by tutors across the disciplines, relate to the broader skills of critical thinking (Wolf *et al.* 2010). The ability to analyse, interpret, infer, explain and evaluate an ethical issue draws upon broader critical thinking skills (Facione 2000). Although the programme outcomes rarely mentioned ethics explicitly, all of them identify these broader skills. Table 5.1 explores these different abilities, highlighting how they are identified in the programme outcomes or programme graduate characteristics. With the exception of the 'Knowledge and Understanding' of the prescribed mode of appropriate behaviour for looking after animals, the other abilities are related to cognitive, key or transferable skills. The findings from the MEQ demonstrate that these abilities are significantly weaker elements of the students' ethical development. As the ability to think ethically is not presently a programme outcome or a programme graduate characteristic, but is recognised as a university level graduate attribute, this suggests that more needs to be done at the programme level to support student ethical development.

Expected ethical	Programme o	Student		
thinking	(P	development		
(interview analysis)	English	(MEQ)		
Ability to recognise	Cognitive skill: Stu	dents will have deve	eloped a grasp of the	Nature of
complexity and to	ways in which m	eaning is produced a	and of the aesthetics	multiplicity
reflect upon issues	of reception.			lowest mean
from different	Transferable skill:	score (12.75)		
perspectives	derived from abs			
	sources to practi			
	capacity to inter			
	claims and argun			
	between alterna			
	Graduate characte			
	and reflective wi	th a high level of ski	ll in problem-solving,	
	project managen	nent, IT and multime	edia skills.	

 Table 5.1: Expected ethical thinking in relation to programme outcomes and student

 development

	 Key skill: Demonstrate an ability to identify, evaluate and propose answers/solutions to complex problems. Cognitive skill: Abstract, synthesise and critically evaluate information from a wide range of sources. Cognitive skill: Bring a critically and theoretically informed perspective to relevant issues and current developments (as appropriate) in the study of animal behaviour and welfare. Graduate characteristic: The ability to sift evidence and evaluate arguments. 				
Ability to reflect	Transferable skill: Students will develop skills in: problem-	Personal			
upon issues	solving.	responsibility –			
independently	Transferable skill: Students will also develop a range of life	second lowest			
	Skills.	mean score			
	Practical skill: Undertake field-based research in an				
	ethical and health and safety issues				
	Key skill: Demonstrate an autonomous and confident				
	approach to study				
	independence in relation to learning				
Ability to learn	N/A	A lack of			
laws and	Ν/Δ	difference			
of behaviour		disciplines in			
	Knowledge and understanding: Demonstrate an	MEQ suggest			
	understanding of key ideas, especially in the following	learning			
	areas: animal welfare, health and disease, research	prescribed			
		modes of			
		behaviour did			
		not influence			
		development			

This chapter set out to analyse tutor expectations of student ethical thinking in the three academic programmes. It has found that there are similarities in the nature of ethical thinking required in the disciplines. These include three main abilities: 1) to empathise with others and recognise different perspectives on issues; 2) to recognise the complexity of issues; and 3) to take responsibility and consider issues independently. Two points are significant from this: 1) these abilities are all broader than just 'ethical' issues, they relate to critical thinking and higher level skills more generally; 2) these abilities, as assessed by the questionnaire, were found to be the weakest for the students across all three disciplines. The final

chapter concludes this dissertation and reflects on all of the findings to discuss the implications for teaching and learning.

Conclusions

In conclusion, this chapter summarises the findings from the first four research

objectives:

- 1. To analyse the ethical development of students in three academic programmes in the arts, social and pure sciences.
- 2. To investigate the extent to which there is progression in the ethical development of students in different years across the three academic programmes.
- 3. To examine the ethical development of students' ethical thinking by gender across the three academic programmes.
- 4. To analyse how the nature of ethical thinking expected by tutors varies between disciplines and evaluate the extent to which this aligns with the students' ethical development.

This is followed by a discussion of the implications of these findings by addressing the final objective of the research:

5. To discuss the implications of the findings for enhancing the teaching and learning of ethics.

The research found that there were no significant differences between disciplines in terms of student meta-ethical development. There was some evidence of differences between years, but not clear evidence of progression over the three years of the undergraduate programme. The main differences found were between male and female students, with male students demonstrating less ethical development than their female counterparts. The research with the tutors found there to be few differences between disciplines in terms of the nature of ethical thinking expected; these ethical skills were strongly related to generic higher education skills. Reflecting on the findings from the tutors and students, this dissertation has highlighted how tutors' expectations of the type ethical thinking of students do not correlate with the development demonstrated by the students in the meta-ethical questionnaire (MEQ). The thinking tutors expected were the weakest elements in the MEQ findings.

6.1 Implications of findings for teaching and learning

The mismatch between the tutor expectations of student ethical thinking abilities and the findings of their students' meta-ethical development indicate the need to reflect on the teaching and learning strategies in place for teaching ethical thinking. As argued earlier, the ability to think through ethical issues has a symbiotic relationship with the ability to think critically (Boni & Lozano 2007). Designing effective ways to support the development of these abilities generally are also likely to support the meta-ethical development of students at the same time (Clarkeburn *et al.* 2003). Two related concepts which may offer useful insights into designing effective teaching and learning strategies are troublesome knowledge and threshold concepts.

Troublesome knowledge is the idea that students find certain types of knowledge particularly challenging (Meyer & Land 2003). Perkins (1999) describes troublesome knowledge as that which appears counter-intuitive, alien, or incoherent. This notion of 'alien' knowledge may help to interpret student difficulties with ethics. Alien knowledge is understood as that which emanates from another culture or discourse (Meyer & Land 2003). As demonstrated by this research students struggled to recognise multiple perspectives on issues. This

relates to challenges of understanding different ways of seeing the world i.e. different discourses.

The notion of threshold concepts offers further insights. A threshold concept is transformative, probably irreversible, integrative, often troublesome and often disciplinarily 'bounded' (Irvine & Carmichael 2009). Meyer & Land (2003) have associated the concept with going through a portal, irrevocably transforming the student's understanding. Although ethical thinking is a skill rather than a concept, and is not, as we have seen, disciplinary 'bounded', elements from the theory of threshold concept may still offer useful perspectives in two ways: 1) the element of integration, and 2) the notion of liminality. Davies (as cited by Meyer & Land 2003) discusses the integrative characteristic of threshold concepts in relation to troublesome knowledge:

"'Integration' is troublesome because you need to acquire the bits before you can integrate, but once you've got the bits you need to be persuaded to see them in a different way" (p. 6).

Focusing on the complexity of ethical situations, this characteristic of a threshold concept, illustrates how in terms of thinking ethically, students need to integrate information from a range of different perspectives. This is troublesome for students as the complexity of the different kinds of information students are dealing with as they assess ethical issues are inconsistent, paradoxical in nature, or may contain subtle distinctions (Meyer & Land 2003). Secondly, the characteristic of liminality highlights student challenges with ethical thinking. Meyer & Land (2003) explain how

"difficulty in understanding threshold concepts may leave the learner in a state of liminality (Latin limen – 'threshold'), a suspended state in which understanding approximates to a kind of mimicry or lack of authenticity" (p. 10).

When it comes to ethical development the suspended state relates to the point at which students are on the edge of grasping the nature of multiple perspectives and complexity.

These insights have implications for devising effective learning and teaching strategies. When dealing with troublesome knowledge and threshold concepts it is important to engage with students in a way which will most benefit the learner in terms of achieving understanding (Hall 2010). Several of these approaches cast the learner in a role as: "active learners (where knowledge and understanding are actively acquired); social learners (where knowledge and understanding are socially constructed) and creative learners (where knowledge and understanding are created or recreated)" (Phillips 1995; cited Hall 2010: 49). As active learners, students play an active role in their learning, moving beyond listening, reading and working through exercises to discuss, debate, hypothesise, investigate and take viewpoints (Perkins 2006). Active learning through activities which encourage students to discuss and debate issues from different perspectives offers potential to support students in understanding how people have different views on ethical issues. Constructivists often emphasise how knowledge and understanding are socially produced (Perkins 2006). Discussing ethical issues with others and recognising that there are multiple perspectives on issues will help to support students to understand how 'truth' varies with interest groups. Finally, the

opportunity for students to create or recreate knowledge may support student learning (Perkins 2006). For example, rather than debating and discussing ethical scenarios, students could create their own ethical scenarios for discussion within the group. A combination of these approaches may help to support students in the key areas where they appear to struggle most: recognising multiplicity and complexity.

One example of a learning strategy that could bring together these three different 'roles' for the learner is debate. The author has argued elsewhere how debate can support the development of critical thinking skills (Healey 2012). Debate may be facilitated in a variety of different formats including role play (Tyrell 2010), panel debates (Green & Klug 1990), and debate between two students (Healey 2012). This may be a 'social learning experience' as students work together to prepare for and 'compete' in the debate; an 'active learning experience' as students learn through the activity of the debate; and a 'creative learning experience' as students design the focus of the debate, decide on the materials to be used and construct question(s) for the debate.

Finally, depending upon whether ethical decision making is seen as a graduate or programme specific skill influences how ethics might be included in the undergraduate curriculum. As a graduate skill linked to employability it might be taught centrally within institutions, bringing students together from a range of disciplines as a separate optional session. Whereas if it is understood as a programme specific skill there are two ways it might be addressed: 1) As a separate

module within a disciplinary programme, focusing specifically on ethics thereby raising student awareness of ethics directly. 2) Embedded into existing modules focusing upon the ethical issues underlying existing discussions. This research suggests that ethical thinking should be a specific programme outcome.

The relevance of ethics to each of the disciplines discussed here has been highlighted, yet this research has also demonstrated the nuances and disciplinary specificity of the discussions (Jenkins 1996; Humber & Morreale 2002). It is essential that in reconsidering the approach to teaching ethics that these disciplinary nuances are not lost, as this is where the main interest lies for the students (Healey 2000; Valentine 2005; Pace & Middendorf 2004). As for whether ethics should be taught as an additional area or embedded within current content, the discussions with tutors highlighted that there are positives and negatives to both approaches. As an additional module, ethics would be explicit to students, enabling focused analysis and discussion of ethical issues. However, by segregating it from the disciplinary content students may compartmentalise their learning and envisage ethics as an 'add-on' to their studies rather than a fundamental element. By embedding ethics into existing teaching some of these issues are overcome. Here ethics is made a part of the discussion of general content. Yet, as has been demonstrated in this research, there is a danger that ethics can be so embedded that students are unaware that they are studying ethical issues. They understand such problems as disciplinary issues, rather than recognising them to be ethical. A sensible strategy to adopt would be to make ethics more explicit within current content, without undermining the main approach. By embedding ethics through

active, social and creative learning within current disciplinary content, students have the opportunity to learn that ethics is part of the discipline.

6.2 Limitations and reflections

This research had three main limitations. Firstly, the meta-ethical questionnaire (MEQ) itself was limited as it assessed a particular type of ethical development, measuring meta-ethics, as opposed to normative or applied ethics. Students may have indicated their response in relation to where they think the research wanted them to be, rather than where they actually believed they were. However, in order to be comparable with Clarkeburn *et al.*'s (2003) study it was necessary to use this same tool. The second limitation relates to the first. By only using a questionnaire to assess student's ethical development it was not possible to follow up with students why they had answered as they had. Time limitations prevented further discussion with the students, as did the issue of how comfortable students would feel discussing ethics one on one. Finally, as the research took place with three independent year groups it was not possible to convincingly make conclusions as to progression during degrees, as the findings may relate to cohort differences. A longitudinal study over multiple years would offer data which could address this.

6.3 Future research

This research adopted a comparative case study approach. This means that the findings are specific to these cases studies, and as such may not be generalizable to other cohorts or universities (Cousin 2005; 2009). That said, the findings raise

interesting questions for further research in other institutions, disciplines and with

different cohorts in the case study disciplines at this University (Table 6.1).

Table 6.1:	Further	research	questions
------------	---------	----------	-----------

	Questions
Institutions	 Do English, Geography, and Animal Behaviour and Welfare receive similar ethical scores in different institutions?
	 If not, what are the differences between a) the students, b) the nature of the way they are taught?
Disciplines	• Do the findings alter if the specific subjects chosen to illustrate the arts, social and pure sciences are changed?
	• If so, what are the differences between a) the students, b) the nature of the way they are taught?
Cohorts	• Do the findings at the case study university alter with different cohorts?
	 Do the findings at the case study university change as the original cohorts' progress through their university studies?
Teaching methods	• How is ethics taught in different disciplines, different institutions and different countries?
	 Is ethical thinking taught differently at postgraduate level?
	 What impact do particular teaching methods have on ethical thinking and development?
Qualitative	 How is ethics understood in different disciplines?
methods	• What do students believe influences the way they understand ethics the way they do?

Understanding how best to support the meta-ethical development of students and ethical thinking skills is becoming increasingly important in response to recent political and financial scandals (Gao *et al.* 2008; Ruhe & Lee 2008; Carrell 2009). With the introduction of up to £9,000 fees for higher education in the UK, students are even more likely to consider their prospects in the graduate employment market (Foskett *et al.* 2006). Institutions and programmes which provide students with an awareness and understanding of ethics within their own discipline offer the opportunity for graduates to stand out from the crowd. Indeed, Bauman (1993: 16-17) would argue that:

"Going about our daily affairs ... we need moral knowledge and skills more often, and more poignantly, than either knowledge of the 'laws of nature' or technical skills".

- Barnett, R. (2000) Supercomplexity and the curriculum, *Studies in Higher Education* 25(3): 255-265.
- Barnett, T. & Brown, G. (1994) The ethical judgements of college students regarding business issues, *Journal of Education for Business* 69: 333-339.
- Barrie, S (2004) A research-based approach to generic attributes policy, *Higher Education Research and Development* 23 (3): 261-275.
- Barrie, S. (2006) Understanding what we mean by the generic attributes of graduates, *Higher Education* 51: 215-241.
- Barrie, S. (2007) A conceptual framework for the teaching and learning of generic graduate attributes, *Studies in Higher Education* 32 (4): 439-458.
- Bassey, M. (1999) Case Study Research in Educational Settings (Milton Keynes: OU Press).
- Bartell, M. (2003) Internationalization of universities: A university culture-based framework, *Higher Education*, 45, 43–70.
- Bauman, Z. (1993) Postmodern Ethics (Oxford: Blackwell).
- Beck, L.G. & Murphy, J. (1994) *Ethics in Educational Leadership Programmes: An Expanding Role* (Thousand Oaks, C.A.: Corwin Press).
- Boni, A. & Lozano, J.F. (2007) The generic competences: an opportunity for ethical learning in the European convergence in higher education, *Higher Education* 54: 819-831.
- Boyd, W. Healey, R.L., Hardwick, S.W., Haigh, M. with Klein, P., Doran, B., Trafford, J. Y Bradbeer, J. (2008) 'None of Us Sets Out To Hurt People': The Ethical Geographer and Geography Curricula in Higher Education, *Journal of Geography in Higher Education*, Vol. 32, No. 1, 37–50.
- Bruhn, J.G. (2008) Value dissonance and ethics failure in academia: a casual connection? *Journal of Academic Ethics* 6: 17-32.
- Carrell, S. (2009) MPs' expenses scandal has dealt devastating blow to politics, says Gove, *The Guardian*, 15th May.
- Clarkeburn, H., Downie, J. R. & Matthew, B. (2002) Impact of an ethics programme in a life sciences curriculum, *Teaching in Higher Education* 1: 65-79.
- Clarkeburn, H.M., Downie, J. R., Gray, C. & Matthew, R.G.S. (2003) Measuring ethical development in life sciences students: a study using Perry's developmental model, *Studies in Higher Education* 28(4): 443-456.
- Cohen, L., Manion, L. & Morrison, K. (2000) *Research Methods in Education*, 5th Edition (London: RoutledgeFalmer).
- Colnerud, G. & Rosander, M. (2009) Academic dishonesty, ethical norms and learning, Assessment & Evaluation in Higher Education 34(5): 505-517.

- Conner, M. G. (1999) Understanding the Difference Between Men and Women, available at: <u>http://www.oregoncounseling.org/ArticlesPapers/Documents/DifferencesMe</u> <u>nWomen.htm</u> [accessed 31st July 2012].
- Cortese, A.D. (2003) The critical role of higher education in creating a sustainable future, *Planning for Higher Education*, March-May: 15-22.
- Cousin, G. (2005) Case study research, *Journal of Geography in Higher Education* 29(3): 421-427.
- Cousin, G. (2009) Researching Learning in Higher Education: An Introduction to Contemporary Methods and Approaches (London: Routledge).
- Dellaportas, S. (2002) The Effect of Accounting Education on Students' Moral Reasoning and Development (Unpublished PhD Theses, RMIT University)
- De Vaus, D. (1991) *Surveys in Social Research*, 3rd Edition (London: Routledge).
- Dexter, L.A. (1970) *Elite and specialized interviewing* (Evanston: Northwestern University Press).
- Donoho, C., Heinze, T., & Kondo, C. (2012) Gender differences in personal selling ethics evaluations: do they exist and what does their existence mean for teaching sales ethics, *Journal of Marketing Education* (34:1) 55-66.
- Ellery, K. (2007) Undergraduate plagiarism: a pedagogical perspective, Assessment & Evaluation in Higher Education 33(5): 507-516.
- Escámez, J., López, R. G., Y Jover, G. (2008) Restructuring university degree programmes: a new opportunity for ethics education? *Journal of Moral Education* 37(1): 41-53.
- Facione, P.A. (2000) The disposition toward critical thinking: its character, measurement and relationship to critical thinking skill, *Informal Logic* 20(1): 61-84.
- Fink, A. (1995) The Survey Handbook (SAGE: London).
- Foskett, N., Roberts, D., & Maringe, F. (2006) Changing Fee Regimes and their Impact on Student Attitudes to Higher Education, available at: <u>http://www.heacademy.ac.uk/assets/documents/research/changing fees re</u> <u>gimes full report.pdf</u> [accessed 6th August 2012].
- Gao, T, Siegel, Pl, Johar, J.S. & Sirgy, M.J. (2008) A survey of management educators' perceptions of unethical faculty behaviour, *Journal of Academic Ethics* 6: 129-152.
- Goudge, P. (2003) *The Whiteness of Power: Racism in Third World Development Aid* (London: Lawrence and Wishart).
- Green, C.S. & Klug, H. (1990) Teaching critical thinking and writing through debates: an experimental evaluation, *Teaching Sociology* 18: 462-471.
- Hall, B.M. (2010) *Teaching Uncertainty: The Case of Climate Change* (Unpublished PhD Thesis, University of Gloucestershire).

- Hargreaves, J. (2008) Risk: the ethics of a creative curriculum, *Innovations in Education and Teaching International* 45(3): 227-234.
- Hay, I. (1998) Making moral imaginations: research ethics, pedagogy, and professional human geography, *Ethics Place and Environment* 1(1): 55–76.
- Hay, I. & Foley, P. (1998) Ethics, geography and responsible citizenship, *Journal of Geography in Higher Education* 22(2): 169–183.
- Hay, I. (2003) Ethical practice in geographical research. In N. Clifford and G. Valentine (Eds.) *Key Methods in Geography* (London: Sage): 37-53.
- Healey, M. (2000) Developing the scholarship of teaching in higher education: a discipline-based approach, *Higher Education Research and Development* 19(2): 169-189.
- Healey, R.L. (2012) The power of debates? Reflections on their potential for geography in higher education: teaching for social transformation through debate, *Journal of Geography in Higher Education* 36(2): 239-257.
- Healey, R.L., Ribchester, C. & Ross, K. (2011) 'The Ethical Student': Enhancing the Teaching of Ethics in the Undergraduate Curriculum, for the Learning and Teaching Institute, University of Chester.
- Higher Education Academy (2006) *The UK Professional Standards Framework for Teaching and Supporting Learning in Higher Education* (York: HEA).
- Higher Education Academy (HEA) (2011) English Language Subject BenchmarkStatement,HEAEnglishSubjectCentrehttp://www.english.heacademy.ac.uk/explore/resources/language/docs/ELbhttp://www.english.heacademy.ac.uk/explore/resources/language/docs/ELbhttp://www.english.heacademy.ac.uk/explore/resources/language/docs/ELbhttp://www.english.heacademy.ac.uk/explore/resources/language/docs/ELenchmarking final.pdf [accessed 10th August 2012]
- Hilder, M.B. (2005) Teaching literature as an ethic of care, *Teaching Education* 16(1): 41-50.
- Hoggart, K., Lees, L. and Davies, A. (2002) *Researching Human Geography* (London: Oxford University Press).
- Humber, M. T. & Morreale, S. (Eds.) (2002) *Disciplinary Styles in the Scholarship of Teaching and Learning* (Carnegie Foundation for the Advancement of Teaching: Menlo Park).
- Jenkins, A. (1996) Discipline-based educational development. *The International Journal for Academic Development* 1(1): 50-62.
- Irvine, N. & Carmichael, P. (2009) Threshold concepts: a point of focus for practitioner research, *Active Learning in Higher Education* 10: 103-119.
- Israel, M. & Hay, I. (2006) Research Ethics for Social Scientists (London: Sage).
- Kelger, D.E. (2011) Gender Differences in Perceptions of Ethics Among Teacher Education Students at San Diego State University (Unpublished PhD Thesis, Alliant International University, San Diego).
- Knapper, C.K. & Cropley, A.J. (2000) *Lifelong Learning in Higher Education*, 3rd Edition (London: Kogan Page Limited).

- Kuh, G.D. (1994) Student Learning Outside the Classroom: Transcending Artificial Boundaries (Washington D.C.: ERIC Clearhouse on Higher Education).
- Land, R., Meyer, J.H.F. & Smith, J. (Eds.) (2008) *Threshold Concepts within the Disciplines* (Sense Publishers: Rotterdam).
- Lane, M.S., & Schaupp, D. (1989) Ethics in education: a comparative study, *Journal* of Business Ethics 8: 943-949.
- Ludlum, M. & Mascaloinov, S. (2004) Right and wrong and cultural diversity: replication of the 2002 NAS/Zogby poll on business ethics, *Journal of Education for Business* 79(5): 294-298.
- Macfarlane, B. & Cheug, M. (2008) Communism, universalism and disinterestedness: re-examining contemporary support among academics for Merton's scientific norms, *Journal of Academic Ethics* 6: 67-78.
- Matthews, D. (2012) Learning to do the right thing and swear to it, *Times Higher Education* 2nd August: 20-21.
- Meyer, J. & Land, R. (2003) Threshold Concepts and Troublesome Knowledge: Linkages to Ways of Thinking and Practising within the Disciplines (Enhancing Teaching-Learning Environments in Undergraduate Courses, Occasional Report 4: Edinburgh), available at: <u>http://www.etl.tla.ed.ac.uk/docs/ETLreport4.pdf</u> accessed 11th August 2012].
- Miller, K.W. (1999) *Teaching Computer Ethics Using the World Wide Web*, Frontiers in Education Conference, 29th Annual Conference.
- Miori, V.M., Doyle, K.A., & Campbell, K. (2011) Gender differences in academic ethics with recommendations for curricular change, *Journal of College Teaching and Learning* 8(3): 7-20.
- Moon, J. (2008) Critical Thinking: An Exploration of Theory and Practice (Abingdon: Routledge).
- National Association of Writers in Education (NAWE) (2008) *Creative Writing Subject Benchmark Statement*, available at: <u>http://www.nawe.co.uk/Private/17547/Live/CW%20Benchmarks.pdf</u> [accessed 10th August 2012].
- Nicholson, C. & DeMoss, M. (2009) Teaching ethics and social responsibility: an evaluation of undergraduate business education at the discipline level, *Journal of Education for Business*, March/April: 213-218.
- Pace, D. & Middendorf, J. (Eds.) (2004) *Decoding the Disciplines: Helping Students Learn Disciplinary Ways of Thinking* (New Directions for Teaching and Learning No 98 San Francisco: Jossey-Bass).
- Perkins, D. (1999) The many faces of constructivism, *Educational Leadership* 57(3): 6-11.
- Perkins, D. (2006) Constructivism and troublesome knowledge in, Meyer, J.H.F. & Land, R. (Eds.) Overcoming Barriers to Student Learning: Threshold Concepts and Troublesome Knowledge (London: Routledge): 33-47.

- Perry, W.G. Jr. (1999) Forms of Ethical and Intellectual Development in the College Years: A Scheme (San Francisco, CA: Jossey-Bass).
- Persons, O. (2009) Using a corporate code of ethics to assess students' ethicality: implications for business education, *Journal of Education for Business* 84(6): 357-366.
- Peterson, D., Rhoads, A., & Vaught, B.C. (2001) Ethical beliefs of business professionals: a study of gender, age and external factors, *Journal of Business Ethics* 31: 225-232.
- Phau, I. & Kea, H.P. (2007) Attitudes of university students towards business ethics: A cross-national investigation of Australia, Singapore and Hong Kong, *Journal of Business Ethics* 72: 61-75.
- QAA (2007a) English (Mansfield: Quality Assurance Agency), available at: <u>http://www.qaa.ac.uk/Publications/InformationAndGuidance/Documents/En</u> <u>glish07.pdf</u> [accessed 29th August 2012].
- QAA (2007b) *Geography* (Mansfield: Quality Assurance Agency), available at: <u>http://www.qaa.ac.uk/Publications/InformationAndGuidance/Documents/ge</u> <u>ography.pdf</u> [accessed 29th August 2012].
- QAA (2007c) *Biosciences* (Mansfield: Quality Assurance Agency), available at: <u>http://www.qaa.ac.uk/Publications/InformationAndGuidance/Documents/Bio</u> <u>sciences07.pdf</u> [accessed 29th August 2012].
- Reiman, A. J. (2002) A comparison of four-year longitudinal studies of postconventional moral judgement reasoning in teacher education and other selected undergraduate samples, *American Educational Research Association Conference*, New Orleans, 5th April.
- Reiss, M.J. (1999) Teaching ethics in science, *Studies in Science Education* 34: 115-140.
- Robinson, S. (2005) Ethics and employability, in: *Learning and Employability* (York: Higher Education Academy): 1–25, available at: <u>http://www.heacademy.ac.uk/assets/documents/employability/ESECT Ethics</u> <u>and employability.pdf</u> [accessed 10th August 2012].
- Rooy, W.V. & Pollard, I. (2002) Teaching and learning about bioscience ethics with undergraduates, *Education for Health* 15(3): 381-385.
- Ruhe, J. & Lee, M. (2008) Teaching ethics in international business courses: the impacts of religions, *Journal of Teaching International Business* 19(4): 362-388.
- Solbrekke, T.D. & Karseth, B. (2006) Professional responsibility an issue for higher education? *Higher Education* 52: 95-119.
- Smith, D. M. (1995) Moral teaching in geography, *Journal of Geography in Higher Education* 19(3): 271-283.

- Smith, S. (2001) Doing qualitative research: from interpretation to action, in Limb,
 M. & Dwyer, C. (Eds.) *Qualitative Methodologies for Geographers: issues and debates*, (Arnold: London): 23-40.
- Tse, A.C.B. & Au, A.K.M. (1997) Are New Zealand business students more unethical than non-business students? *Journal of Business Ethics* 16(4): 445-450.
- Tyrell, N. (2010) Personal communication. 20th July 2010.
- University [in England] (2010) *The University Mission*, available at: <u>http://ganymede.ac.uk/index.php?page_id=1330463</u> [accessed 10th August 2012].
- Valentine, G. (2005) Tell me about...: using interviews as a research methodology, in R. Flowerdew and D. Martin (Eds.) *Methods in Human Geography: A Guide For Students Doing a Research Project*, 2nd Edition (Harlow: Longman): 110-127.
- Valentine, R. (2005) Because Hester Prynne was and existentialist, or why using disciplines as frameworks for learning clarifies life, Riordan, T. & Roth, J. (Eds.) Disciplines as Frameworks for Student Learning: Teaching the Practice of Disciplines (Virginia, USA: Stylus Publishing): 123-134.
- Vujakovic, P. & Bullard, J. (2001) The ethics minefield: issues of responsibility in learning and research, *Journal of Geography in Higher Education*, 25(2): 275– 283.
- Walesh, S.G. (2012) Engineering Your Future: The Professional Practice of Engineering, 3rd Edition (New Jersey: Wiley & Sons).
- Watson, D. (2007) Does higher education need a Hippocratic oath? *Higher Education Quarterly* 61(3): 362-274.
- Wellens, J., Beradi, A., Chalkley, B., Chambers, B., Healey, R.L., Monk, J. & Vender, J. (2006) Teaching geography for social transformation, *Journal of Geography in Higher Education* 30(1): 117-131.
- Whipple, L.P. & Swords, D.F. (1992) Business ethics judgements: a cross-cultural analysis, *Journal of Business Ethics* 11: 671-678.
- Wolf, J., Stanton, M. & Gellott, L. (2010) Critical thinking in physical geography: linking concepts of content and applicability, *Journal of Geography* 109: 43-53.
- Yin, R. K. (2002) Case Study Research, Design and Methods, 3rd Edition (Newbury Park: Sage).

Ethical thinking in a the disciplinary context

This short questionnaire is part of a teaching and learning project being conducted in the Learning and Teaching Institute. The project aims to analyse how different disciplines teach and understand ethics. You are invited to complete this questionnaire, as your views would be very helpful to our research and will be used to help us understand how to support your learning. More information about the project is available on the Participant Information Sheet at the end of this questionnaire. An electronic version of this questionnaire is available for download at: <a href="http://ganymede.com/http://g

Your completion of this questionnaire is entirely voluntary. The information collected will be anonymous.

Information about you

Please tick next to the relevant information about yourself:

Gender:	Male []	Age:	<18 []	Year:	Level 4 []		
	Female []		18[]		Level 5 []		
			19[]		Level 6 []		
			20 []				
			21[]				
			22-25 []				
			26-30 []				
			31-40 []				
			41-50 []				
			51-60 []				
			>60 []				
Programme	e of study:						
[] Single Hon	ours English						
[] Combined	Honours English AND (please indicate)					
[] Single Honours Animal Behaviour and Welfare							
[] Single Honours Animal Behaviour							
[] Combined Honours Animal Behaviour AND (please indicate)							

What does the term 'ethics' mean to you?

For example, what do you think of when you hear the term?

In your previous studies at university or school, please provide an example of an ethical issue you have studied:

Please provide an example of an ethical choice that you have faced or made as part of your school or university related activities:

Your Thoughts On Ethics

Please complete the following by putting an X in the box which represents the statement most close to your opinion <u>(one X per line)</u>. Please be honest about your views, there are no right or wrong answers, where you position yourself in relation to the statements reflects your personal perspectives on different issues.

	Definitely my opinion	More or less what I believe	Neither statement represents my view	More or less what I believe	Definitely my opinion	
Moral questions have absolutely right answers.						There are very few absolutely right answers in the world and answers to moral questions are not one of them.
Personal moral values are the same forever.						Personal moral values need to be reconsidered from time to time.
People cannot choose their values because values are either right or wrong.						I am committed to a set of values I have chosen for myself.
I do not doubt that my values are the right values to have.						I need to commit myself to a set of values even when I am uncertain whether they will always be the right values to have.
It is not my place to make moral choices, because right answers have been found already by others.						When I have a moral problem I try to think the answer through myself.
A good moral answer is short and simple, because you know the right answer.						You cannot have a good moral answer without arguments to support it, because moral answers are never straight forward.
I believe we can always make a judgement whether actions are right or wrong and these rules do not change.						When we make moral decisions, the best we can do is to decide what is right as far as we can tell in different situations.

I don't think discussing moral problems is beneficial for me unless a right answer can be found at the end.			Discussing values with other people gives me a beneficial opportunity to reflect on my values, even when there is no agreement in the end.
I don't enjoy discussing moral problems unless the teacher can give the right answer in the end.			I enjoy discussing my values in the class even when we cannot agree on one right answer in the end.
I don't think teachers should assess my moral arguments if they do not know the right answers yet.			It is important that teachers assessing moral arguments look for logical structure and good reasoning rather than a particular answer.

Source: Clarkeburn, H.M., Downie, J. R., Gray, C. & Matthew, R.G.S. (2003) Measuring ethical development in life sciences students: a study using Perry's developmental model, *Studies in Higher Education* 28(4): 443-456.

Participant Information Sheet: Ethical thinking in a disciplinary context Questionnaire

You are being invited to take part in a research study. Before you decide, it is important for you to understand why the research is being done and what it will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask me or the tutor who gave you this questionnaire if there is anything that is not clear or if you would like more information. Take time to decide whether or not you wish to take part.

What is the purpose of the study?

This research will inform the development of how ethics is taught within the University by exploring student understandings of ethics at different stages in their degree programmes. The first stage of this research is to gather baseline information by use of a questionnaire. The purpose of this questionnaire is to explore your views of ethics and your own ethical development.

Why have I been chosen?

All the students on 5 different programmes are being invited to answer this questionnaire. These are: Single and Combined Honours English; Single and Combined Honours Animal Behaviour; and Single Honours Animal Behaviour and Welfare.

Do I have to take part?

Completion of the questionnaire is entirely voluntary. Your decision to participate or not will not provide any advantage or disadvantage to you. We would, however, greatly appreciate you taking the time to complete the questionnaire.

What will happen to me if I agree to take part?

This is a paper-based questionnaire which students in all year groups will be asked to complete at the beginning of the year in class (or in their own-time if they are unable to attend the class and wish to participate). It is expected that the questionnaire should take no longer than 15 minutes to complete. The questionnaire requires you to consider ethical choices or decisions you have made in the past and your reasoning behind your decisions. Your completion of the questionnaire will be taken as your consent to participate in the research.

Why do you ask me what programme I am studying?

We are trying to understand the influence of different programmes upon ethical understanding and ethical development. As such it is important to know which programme you are studying. However, if you do not wish to identify your programme, you do not have to.

Who is organising and funding the research?

This research is part of a Masters Dissertation within the discussion of a Masters Dissertation of a Masters Dissertation within the discussion of a Masters Dissertation of a Mas

What will happen to the results of the research study?

The results will be analysed by the researcher detailed below. When any results and findings of this research project are presented or reported to others inside or outside of the University, your anonymity is guaranteed. Reference to specific people, who you may mention, will also be removed from any quotations that are used.

Will my taking part in the study be kept confidential?

Yes. To ensure this please <u>do not write your name on the questionnaire</u>.

What if something goes wrong?

We recognise that sometimes recalling ethical choices and decisions may cause distress. If this is the case, and you wish to talk about this to anyone, please contact Student Counselling (Tel: 01244 511550 or <u>E-mail:</u> <u>student.counselling@_____ac.uk</u>). If you wish to complain or have any concerns about any aspect of the way you have been approached or treated during the course of this research, please contact Professor Jethro Newton, Dean of Academic Quality and Enhancement, <u>Courselling</u>, Email: <u>i.newton@___ac.uk</u>, Tel: 01244 511938.

Who may I contact for further information?

Dr. Ruth Healey r.healey@chester.ac.uk 01244 513176

Thank you for your interest in this research.

Participant Information Sheet: Ethical thinking in a disciplinary context Interviews with Tutors

You are being invited to take part in a research study. Before you decide to take part, it is important for you to understand why the research is being done and what it will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask me if there is anything that is not clear or if you would like more information. Take time to decide whether or not you wish to take part.

What is the purpose of the study?

This research will inform the development of how ethics is taught within the University by exploring student understandings of ethics at different stages in their degree programmes.

Why have I been chosen?

You have been selected because you are a programme leader or tutor on one of the case study programmes.

Do I have to take part?

Participation in this interview is entirely voluntary. Your decision to participate or not will not provide any advantage or disadvantage to you. I would, however, greatly appreciate you taking the time to participate in the interview.

What will happen to me if I agree to take part?

The interview will involve reflecting upon how ethics is taught in your programme. You will be asked about your perceptions of ethics, including its relative importance, and how you think it would be best to teach ethics to students at different stages in their development. You will be interviewed by Ruth Healey. It is expected that the interview should take no longer than an hour and will be recorded using a digital recorder. You will be asked to sign a consent form.

Who is organising and funding the research?

This research is part of a Masters Dissertation within the **second second second**'s Learning and Teaching Institute.

What will happen to the results of the research study?

The results will be analysed by the researcher, Ruth Healey. When any results and findings of this research project are presented or reported to others inside or outside of the University, your anonymity is guaranteed. Reference to specific people, who you may mention, will also be removed from any quotations that are used.

What if something goes wrong?

We do not expect anything to go wrong but if you wish to complain or have any concerns about any aspect of the way you have been approached or treated during the course of this research, please contact Professor Jethro Newton, Dean of Academic Quality and Enhancement, ______, Email: j.newton@_____ac.uk, Tel: 01244 511938.

Who may I contact for further information?

Dr. Ruth Healey r.healey@chester.ac.uk 01244 513176

Thank you for your interest in this research.

Consent form to participate in interview

Title of Project:

Ethical Thinking in a Disciplinary Context

Names of Researchers:

Dr Ruth Healey

PLEASE INITIAL BOX

- 1. I confirm that I have read and understand the information sheet for the above study and have had the opportunity to ask questions.
- 2. I understand that my participation is voluntary and that I am free to withdraw at any time, without giving any reason or legal rights being affected.
- 3. I am willing for the interview to be recorded.
- 4. I agree to take part in the above study.

Name of Participant

____/___/____ Date

Signature

Researcher

/____/____ Date

Signature



No significant difference by year p < 0.185, Chi-Square 6.199, df4



No significant difference by gender p < 0.261, Chi-Square 4.001, df3



No significant difference by year p < 0.720, Chi-Square 3.678, df6



Significant difference by gender p < 0.051, Chi-Square 7.773, df3



No significant difference by year p < 0.098, Chi-Square 13.437, df8



Significant difference by gender p < 0.045, Chi-Square 9.754, df4



No significant difference by year p < 0.697, Chi-Square 2.210, df4



No significant difference by gender p < 0.550, Chi-Square 1.197, df2



Significant difference by year p < 0.004, Chi-Square 15.274, df4



No significant difference by gender p < 0.315, Chi-Square 2.308, df2